The impact of benefit and tax uprating on incomes and poverty



This publication can be provided in other formats, such as large print, Braille and audio. Please contact: Communications, Joseph Rowntree Foundation, The Homestead, 40 Water End, York YO30 6WP. Tel: 01904 615905. Email: info@jrf.org.uk

The impact of benefit and tax uprating on incomes and poverty

Holly Sutherland, Martin Evans, Ruth Hancock, John Hills and Francesca Zantomio



The **Joseph Rowntree Foundation** has supported this project as part of its programme of research and innovative development projects, which it hopes will be of value to policy-makers, practitioners and service users. The facts presented and views expressed in this report are, however, those of the authors and not necessarily those of the Foundation.

Joseph Rowntree Foundation The Homestead 40 Water End York YO30 6WP Website: www.jrf.org.uk

About the authors

Holly Sutherland is a Research Professor in the Institute for Social and Economic Research (ISER) at the University of Essex.

Martin Evans is Senior Research Fellow in the Department of Social Policy and Social Work at the University of Oxford.

Ruth Hancock is Professor of the Economics of Health and Welfare in the Health Economics Group at the University of East Anglia.

John Hills is Professor of Social Policy and Director of the Centre for Analysis of Social Exclusion at the London School of Economics.

Francesca Zantomio is a Senior Research Officer in the Institute for Social and Economic Research (ISER) at the University of Essex.

© Essex University 2008

First published 2008 by the Joseph Rowntree Foundation

All rights reserved. Reproduction of this report by photocopying or electronic means for non-commercial purposes is permitted. Otherwise, no part of this report may be reproduced, adapted, stored in a retrieval system or transmitted by any means, electronic, mechanical, photocopying, or otherwise without the prior written permission of the Joseph Rowntree Foundation.

ISBN: 978 1 85935 641 8

A CIP catalogue record for this report is available from the British Library.

Prepared and printed by: York Publishing Services Ltd 64 Hallfield Road Layerthorpe York YO31 7ZQ Tel: 01904 430033; Fax: 01904 430868; Website: www.yps-publishing.co.uk

Further copies of this report, or any other JRF publication, can be obtained from the JRF website (www.jrf.org.uk/bookshop/).

Contents

Ac	Acknowledgements v				
Lis	st of abbreviations	viii			
1	Introduction	1			
2	Issues of principle The starting point	4 8			
3	International practice Adjustment mechanisms across the world Nordic countries Australia, Canada and New Zealand New member states of the European Union Summary	12 12 15 16 17 18			
4	Recent practice in the UK	20			
5	Policy in the future, as it looks now	24			
6	Implications of current indexation rules for incomes and poverty The implications of future indexation for a lone parent Implications for other family types Impact on the whole population Budgetary effects Summary	29 32 33 36 40 43			
7	Implications of current indexation rules: work incentives and reliance on means-tested benefits Summary	46 50			
8	Implications of current indexation rules: fiscal drag and benefit erosion The contributions of fiscal drag and benefit erosion Contributions to poverty rates and poverty gaps Contributions to changes in marginal tax rates The effects of 'pure' price indexation Summary	51 52 53 54 57			
9	Implications of alternative indexation strategies Tax 'cuts' Benefit 'increases' Targeted benefit or tax credit increases Summary	59 63 63 64 72			

10 The older population and uprating policy			
The impact of uprating policies on pensioners as they age	77		
The impact of uprating policies on differences in the incomes of current and future pensioners Summary	81 88		
11 Conclusions: choices and trade-offs	90		
Notes			
References			
Appendix 1: Current benefit uprating policies			
Appendix 2: Base case uprating assumptions			
Appendix 3: Structural changes announced in the 2007 Budget and			
accounted for in the 2006–07 starting point system	112		
Appendix 4: Methods and models			

Acknowledgements

We are grateful for useful advice and suggestions from the project Advisory Group members James Arthur, Jonathan Athow, Anna Bee, Mike Brewer, James Browne Chris Drane, Chris Goulden, Donald Hirsch, Archy Kirkwood Mervyn Kohler, David Piachaud, Chris Pond and Julia Sweeney; and also to Fran Bennett, Richard Berthoud, Jonathan Bradshaw and to participants at seminars held in ISER at the University of Essex, at STICERD, LSE and at the OECD. Remaining errors and the opinions expressed remain, however, the responsibility of the authors. We are also grateful to the Department for Work and Pensions for access to the 2003–04 Family Resources Survey, distributed by the Economic and Social Data Service. Crown Copyright material is reproduced with the permission of the Controller of HMSO and the Queen's Printer for Scotland. Martin Evans is an ESRC Research Fellow and acknowledges ESRC support under grant number RES-000-27-0180.

List of abbreviations

AHC	after housing costs
ASHE	Annual Survey of Hours and Earnings
BHC	before housing costs
BSP	Basic State Pension
CPI	Consumer Prices Index
СТВ	Council Tax Benefit
CTC	Child Tax Credit
DWP	Department for Work and Pensions
FIS	Family Income Supplement
FRS	Family Resources Survey
GC	Guarantee Credit
GDP	gross domestic product
HB	Housing Benefit
HBAI	Households Below Average Income
HMRC	HM Revenue & Customs
HRT	Higher Rate Threshold
IS	Income Support
JSA	Jobseeker's Allowance
LOIS	Lifetime Opportunities and Incentives Simulation model
METR	marginal effective tax rate
MTAWE	male total average weekly earnings
MTB	Means-tested benefit
NES	New Earnings Survey
NICs	National Insurance contributions
PC	Pension Credit
RPI	Retail Prices Index
S2P	State Second Pension
SC	Savings Credit
SERPS	State Earnings Related Pension
TBMT	Tax Benefit Model Tables
UB	Unemployment Benefit
UEL	Upper Earnings/Profits Limits
WTC	Working Tax Credit

1 Introduction

Each year in the UK, around 330,000 people reach the age of 80. They then – if entitled to a Basic State Pension in their own right – are told that they will receive an addition of up to 25 pence per week in their state pension. This is often the cause of hilarity about whether to spend the increase on a stamp for a letter of thanks to a munificent government – and possibly also a cause of resentment at such a derisory amount if the Government really thinks that those over 80 have additional needs compared to younger pensioners.

This addition has been 25 pence per week in cash ever since it was introduced in September 1971. Had its value been increased each year in line with the growth of average earnings, it would have been worth nearly \pounds 5 per week in April 2007 – not a large sum, but still significant by comparison with the basic pension of £87.30 per week.

Famously, of course, the previous link between the basic pension itself and earnings was broken in 1981, and since then annual adjustments ('upratings') have been mostly in line with the slower increase in prices. Had the basic pension risen in line with average earnings growth since 1980, it would have been £137 per week in April 2007 – more than half as high again as its actual value.

Or, to take another example, the single personal tax threshold was set at \pounds 595 per year when the income tax system was unified in 1973. Had that value also kept up with average earnings growth, it would have reached \pounds 8,780 per year in 2006–07. The difference between this and the actual threshold of \pounds 5,035 is worth \pounds 825 per year to a basic rate taxpayer in terms of the higher income tax being paid.

As these examples illustrate, what happens over time to the way in which benefits and taxes are uprated can have large implications for the ways in which different people are treated. It also has major implications for public spending. The social security budget in Great Britain was £118 billion in 2006–07. A decision to increase all of its parameters in line with earnings growth in 2007–08 would have meant a cash increase of about £4 billion. If it had all been increased in line with prices, the cash increase in spending would have been only £2.8 billion (with increases in inflation measured by the Retail Prices Index). To put it the other way round, a decision to freeze all the cash values of the system would have meant 2007–08 spending around £4 billion less than a decision to preserve the value of all benefits relative to earnings.

Arguably, by the far the most important social security decision of the Conservative Governments under Mrs Thatcher was to break the earnings links that they inherited in 1979. In the long term, the failure of benefit incomes generally to keep pace with other incomes has been one of the biggest influences on the widening of living standards (Bradshaw and Lynes, 1995; Hills, 2004, pp. 90–3).

Such adjustments – and their feasibility if not made regularly or automatically – can also lead to structural changes in policy. By the time the dog licence was abolished in 1987, it was still set (bar the odd halfpenny) at the 37 pence equivalent of its predecimalisation value of 7s 6d, presumably making it more expensive to collect than the revenue brought in. More seriously, the Government abolished the old taxation of the annual value of owner-occupied property under 'Schedule A' of income tax in 1963, rather than face the consequences of trebling the amounts subject to tax from their 1936 cash values. The system of domestic rates in Great Britain was replaced by the Poll Tax in the 1980s following the political impact of a long-delayed rating revaluation in Scotland. Again, a large part of the recent debate around pension reform has centred on the long-term consequences of the default policies under which the basic state pension was assumed to be uprated each year in line with price inflation, while the means-tested minimum for pensioners (currently the 'Guarantee Credit') would be increased in line with earnings growth.

Uprating rules also have particular relevance at present for the prospects for meeting the current Government's child poverty targets in 2010 and 2020, especially those targets that are set as the proportion of children with household income less than 60 per cent of the contemporary median. If median income rises faster than benefit incomes, then headway towards the targets will be virtually impossible to achieve (Sutherland *et al.*, 2003; Glennerster *et al.*, 2004; Evans and Scarborough, 2006; Hirsch, 2006; Brewer *et al.*, 2007).

What happens each year to the levels of benefits and tax allowances thus has major implications for the standards of living of different groups, for the rate and depth of poverty, and for the public finances. And yet many of the adjustments are made – or not made in cases such as the 25 pence extra pension at 80 – by default and with little debate. Even when the breaking of the earnings link was announced in 1979, the Secretary of State, Patrick Jenkin, said that beneficiaries could still look forward to 'sharing in the increased living standards of the country as a whole' (Bradshaw and Lynes, 1995, p. 15), rather than this being debated as a long-term change in policy towards price uprating. Adjustments also follow a widely varying set of rules (see Chapter 3 of this report). While, as the next chapter discusses, there may be good reasons for following different rules where the objectives of parts of the tax and benefit system vary, it is by no means clear that either the public or politicians understand the long-run implications of these rules.

This report is intended to stimulate debate around these issues. The next chapter discusses the range of principles around the ways benefits, tax credit rates, tax allowances and tax thresholds are set, and the implications of different objectives for how their values might be adjusted from year to year. Chapter 3 discusses the principles used in other countries, while Chapter 4 describes recent practice in the UK. Chapter 5 sets out how the UK system would evolve in coming years if recent conventions continued to be followed, taking account of the structural reforms to the direct tax and tax credit systems announced up to the 2007 Budget. Chapter 6 examines the implications of recent conventions and stated policy regarding uprating (which we term the 'base case'), looking over the short term (six years) and medium term (20 years). The gradual effects of changes over several years are modelled as if they had taken place all at once for a series of hypothetical individuals and for the population as a whole. After considering the effects on income distribution and poverty in Chapter 6, Chapter 7 describes the effect of base case uprating on work incentives and Chapter 8 explores how much of the base case is due to effects on benefits and how much on taxes. As will be seen, the implications of this would be, other things being equal, a very substantial boost to the public finances, with tax revenues rising and benefit spending falling as shares of national income, as well as substantially rising relative poverty. Chapter 9 therefore looks at the implications of alternative scenarios that would involve less of a gain to the public finances and less of an increase in poverty.

In Chapter 10, we look at levels of income for the older population, in particular examining the implications of the reforms introduced by the 2007 Pensions Act by comparison with previous indexation conventions for state pensions. It takes some different perspectives, examining the effects on a cohort of older people as they age through six and 20 years, and following hypothetical individuals through retirement. Chapter 11 concludes.

Although in many ways invisible, and often seen as technical, the manner in which benefit and tax credit levels as well as tax thresholds are adjusted from year to year is one of the biggest decisions in British politics. One can draw a parallel with climate change, albeit on a more parochial level. The effects are only gradual and seem imperceptible on a year-to-year basis. But how we respond to them has immense implications for the future. The aim of this report is not to set out any particular prescription for future decisions, but to make more visible the scale and implications for the income distribution, poverty rates, incentives and public finances of current conventions and of some alternatives.

2 Issues of principle

In the UK, most of the rules determining the ways in which the benefit and tax credit systems are run are set in cash amounts, as are important aspects of the direct tax system. Without adjustment, if there is inflation and real income growth, these amounts will steadily lose value both in terms of what people could purchase and relative to other income sources. This contrasts with the most important parts of the social security systems of some other countries, where, for instance, contributory benefit levels are set as a proportion of past earnings and so are – at initial payment at least – automatically buoyant with those earnings. Making adjustments from year to year – sometimes known as 'bracket indexation' in the context of taxes – is the focus of this report. This is often referred to as 'uprating policy'.¹

Under the heading of 'uprating policy', we might classify the following procedures:

- a formula fixed in advance by law, requiring government intervention to suspend or vary the rule;
- a formula promised in advance but not legally binding;
- a formula that is applied as a convention but is not an advance commitment;
- an ad hoc change in benefit level without reference to a formula.

In this report we consider all but the last of these to be 'uprating'.

In assessing current uprating practice and alternatives, it is helpful to distinguish why particular approaches might be appropriate in different circumstances. These are summarised later in this chapter in Table 1.²

First, if the tax and benefit system is constructed with aims of affecting *inequality in living standards,* then adjustment in line with a measure of mean net (after tax) income growth would be appropriate as a starting point. If this is not done, and benefits are of greater importance for some (usually those with lower incomes) than others, their incomes will fall behind – what we describe below as 'benefit erosion'. Equally, if the brackets or thresholds of the taxes on income (income tax and National Insurance contributions) are not adjusted with income growth, the shares of different people's income taken in tax will change, and with them the shape of the income distribution. Such 'fiscal drag' has implications for both the distribution of the tax burden and the amount of tax revenue collected (Hills and Sutherland, 2004). By

contrast, if all incomes grow by the same proportion, and with them benefit rates and tax brackets, the relationship between net and gross incomes will stay the same, as would direct tax revenues as a share of aggregate income.

Related, but distinct from aims connected with the overall shape of the net income distribution, are aims connected with a specific part of it. In particular, if benefits and tax credits are designed to achieve reductions in *relative poverty*, their values need to rise *at least* with the benchmark of income (or other measure of resources) used to set a poverty standard. In the UK at present, this would mean keeping up with a measure of *typical incomes* (specifically, median household disposable income adjusted for household size). The motivation behind the concept of relative poverty – that resources should be judged relative to the prevailing typical level – could be applied more generally across the income distribution, not simply in relation to the poor.

As a variant of this, some argue that the benefit rates that set society's minimum should be set explicitly to allow those dependent on them to achieve a *minimum living standard* set in relation to contemporary views. Adjusting benefits in line with such 'minimum income standards' requires a three-stage process: periodic recalibration of just what it is that is required for a minimum living standard, finding out what this budget costs to achieve given contemporary prices and, finally, establishing what income is necessary to deliver it. Adjustments to taxes and benefits then follow a combination of the changing minimum (consumption) standards and changes in prices such that disposable incomes keep track with the evolution of the cost of the minimum budget.

By contrast, if benefits are designed to guarantee a particular *real*, but unchanging, standard of living, adjustment would need to be by prices – potentially by a *price index* specific to the group receiving the benefit. For instance, the aim may be to allow a pensioner to continue to purchase the same basket of goods, in which case the relevant price index is for that basket – or for a proxy, such as the 'pensioner prices index'. In the UK case, a special price index – the Rossi index – which excludes changes in housing costs is used to adjust certain social security benefits that are assumed to cover non-housing costs (housing costs being assumed to be covered by the Housing Benefit system). In other cases in the UK, the Retail Prices Index (RPI) for all items is used. An alternative, the Consumer Prices Index (CPI), is commonly used in other countries, although there are also examples of special price indices, such as the Belgian 'health' price index that excludes tobacco and alcohol prices, to which all the main components of the benefit system are linked.³

Fifth, if benefits are seen in some way as a *return on contributions*, as is loosely embodied in the UK national insurance system (but more firmly applied in some other continental European countries), then their values may be linked to what people have paid in or the 'contribution base'. This might suggest a link with, for instance, *gross earnings* (as opposed to net incomes or prices).

Sixth, however, policy may be most driven by questions of *affordability* in terms of the public finances, rather than by what might ideally be achieved. In this case, it may be growth in the economy or *potential tax base* that is most relevant – for instance, gross domestic product (GDP) growth. At times of high growth, benefit levels would rise such that recipients had some share in the country's increasing prosperity. Interestingly, the UK Treasury's long-term fiscal projections assume no fiscal drag and a constant tax to GDP ratio, although benefits are generally assumed to be linked to prices (HM Treasury, 2007a, Annex A).⁴

These aims may be relevant for different aspects of the tax and benefit system, so it is not necessarily illogical for its components to be adjusted in different ways. One instrument may also be contributing to more than one objective, leaving a difficult balance if the relevant reference indices change at different rates. As discussed in the next chapter, some countries use weighted averages of different indices to uprate certain benefits, to some extent reflecting a compromise between multiple goals.

Figure 1 compares how some of the measures discussed here have moved since the early 1970s, showing how the implications of the effects of uprating are very different depending on the index used and the underlying principle guiding its choice. Earnings have grown by nearly twice as much as prices (measured using RPI) over the 36 years shown. Median household disposable income (equivalised as for poverty measurement) has grown at a rate between that of prices and earnings – at 85 per cent of earnings, but 62 per cent more than prices.

Components of the system not only contribute in a number of ways, they also interact. Differential uprating, widely applied in other countries, will alter the nature of the interaction and the people affected by it. For example, a major issue in the recent debate around UK pension reform was the relativity between the Basic State Pension and the means-tested minimum income for pensioners (the Guarantee Credit level). If the former had remained price linked indefinitely, but the latter earnings linked (as in recent policy and assumed by long-term Treasury projections), the proportion of pensioners potentially affected by means testing would have grown substantially over time (Pensions Commission, 2006). Such developments may or may not be in line with the objectives lying behind the differential uprating. So a further principle might be *coherence*. Following this, regardless of the size of the annual adjustment,



Figure 1 Prices, incomes and earnings, 1970–2006 (1970 = 100)

Sources: Retail Prices Index (RPI), Rossi/New Rossi index and Average Earnings Index (DWP, 2006a). Gross domestic product at market prices YBHA: ONS, www.statistics.gov.uk/STATBASE/tsdataset. asp?vlnk=574&More=Y.

Household disposable income (equivalised) mean and median: IFS inequality and poverty spreadsheet accompanying IFS Commentary No. 101, *Poverty and Inequality in Britain: 2006*, downloadable from www.ifs.org.uk/projects_research.php?project_id=127.

it should be common across all parts of the system. As Chapter 3 discusses, some countries use multiples of a common amount (such as the 'guaranteed wage' case in Slovenia or the 'base amount' in Sweden) to specify different parts of their systems, so that adjustment of that common amount affects each of them in proportion.

While one or more principled objectives may underlie uprating policy, what can be done in practice may be constrained by other policy objectives. In particular, concern about work incentives may have the effect of dampening enthusiasm for uprating by more than some minimum. For some objectives, securing a widening gap between incomes in and out of work, or at least not narrowing it, might operate as a constraint on what is feasible in terms of increasing benefits. If benefits rise any less fast than the earnings that benefit recipients might receive, then, other things being equal, the incentive to work may improve. However, as will become clear, and as we explore further in Chapter 7 of this report, what happens to the indexation of in-work benefits and tax credits matters as well as what happens to out-of work benefits. The implications for work incentives of slow rates of indexation of the whole system are not necessarily obvious.

As well as the choice of index, there is a number of technical issues related to the timing of benefit increases and the way they are calculated and rounded that affect the actual benefit payments received. These issues are important, and are discussed further in Box 1, but are not the focus of this report. In our modelling work discussed below, we do not apply rounding and apply constant rates of inflation and earnings growth beyond the immediate future.

The starting point

Setting out such principles takes for granted, however, that the starting levels are what are in fact desired and the only issue is adjustment to keep them in line with the appropriate benchmark for the objective. However, policy-makers are often faced with the challenge of reforming a system from what is, from their point of view, an unsatisfactory starting point. One way of dealing with this is, of course, immediate reform to reach the desired level at once. Often this would mean immediate 'cash losers', as those now seen as overgenerously treated have their weekly benefits cut. It might also mean immediate jumps in tax bills and cuts in take-home pay, as the tax burdens for different groups are rebalanced. While those gaining from such changes would receive their gains at once as well, conventional wisdom is that the political costs from visible losers are usually greater than the political benefits of visible gainers (even if the two groups are evenly matched in size). Either as a way of exploiting 'money illusion' to disguise the losses that are occurring for some, or as a way of slowly phasing in structural changes in a way that minimises disruption to household budgets, transitions to a new system are often made by using more or less generous uprating than implied by the underlying principle, until the intended structure is reached.

For instance, the behaviour of governments since 1971 suggests that they do not, in fact, believe that there is any reason for the basic pension for those aged 80 to be any higher than that for those aged 79. However, rather than risk the opprobrium of simply abolishing the addition one day, they have been content to let it 'wither on the vine' – a tactic that was important, for instance, in the eventual abolition of mortgage

interest tax relief, achieved only when what had originally been a high limit in relation to house prices had fallen far behind them, so the relief had a relatively small impact on people's mortgage payments. In the other direction, desired rates can be achieved from below, perhaps minimising their costs when introduced (and possibly maximising the occasions on which gainers might feel cheerful) – the way in which the National Minimum Wage has increased *faster* than average earnings since it first came in might be an example of this kind.

The difficulty of such adjustment from 'policy disequilibrium' is that what is happening is seldom made explicit – indeed, sometimes the objective is precisely to disguise from some of those affected what is going on. So a final principle that might guide uprating practice might be that of *flexibility* – to minimise the extent of uprating, leaving the Government free to make the adjustments that follow from their political priorities and the economic circumstances, and permitting it to claim credit for increases that in fact maintain only the status quo. Having no rule at all *could* be quite consistent with desirable outcomes if there was full public understanding of the issues at stake and *if* changes in benefits and taxes, as well as redistribution of incomes, were a major item of public discussion and democratic debate.

Indeed, the period since 1997 has seen the Labour Government's goal to reduce child poverty reflected in increases in maximum payments for children through the benefit and tax credit systems that amount to more than they would have with regular uprating by earnings. For example, in 1997, the amount paid for a first child aged under 11 and living with two parents through Income Support was £27.70 per week. If this had been increased in line with earnings, it would have been £41.79 in 2006-07. Following ten years of reform, the combined value of Child Benefit and maximum Child Tax Credit in 2006–07 was in fact £61.87 per week for a first child (if aged over 1) – worth nearly 50 per cent more than uprating by earnings. This restructuring of child payments should not be seen in isolation, however. Not only is the effect of restructuring different for children in one-parent families, by family size and by age of child, the effect on household income and hence the risk of being in poverty depends on other parts of the benefit and tax systems supporting other household members, as well as other factors. For an analysis of the overall effects of the reforms on poverty and inequality since 1997, see Sutherland et al. (2003) and Brewer et al. (2007). The effects relative to the growth in incomes as well as in relative price terms are considered by Hills (2004, Figure 9.5) and Hills and Sutherland (2004).

Table 1 summarises the discussion of this chapter in terms of the links between aims and uprating systems. However, as becomes clear in the review of practice in other countries in the next chapter, it is often not straightforward to distinguish between different objectives, or between uprating and phased structural reform.

Air	n	Implication for choice of uprating factor	
1.	Unchanging inequality	Average income (e.g. mean after tax income)	
2.	Unchanging relative poverty	Typical incomes (e.g. median equivalised household disposable income)	
3.	Keeping track with a Minimum Income Standard	Change in income necessary to achieve a contemporary budget standard	
4.	Unchanging real standard of living	Relevant price index (e.g. Retail Prices Index, Rossi Index or Consumer Price Index)	
5.	Constant return on contributions	Gross earnings	
6.	Public finance affordability	National income (GDP)	
7.	Coherence	Common factor across elements of the tax-benefit system	
8.	Government flexibility	None	

 Table 1 Indexation: summary of aims and implications

Box 1 Rounding and timing issues

Applying exact adjustments in line with percentage amounts of price or income growth could mean amounts that were previously easily expressed amounts - so many pounds per week or hundreds of pounds per year - would become amounts with odd numbers of pence, or even fractions of pence. To ease administration, the results of such calculations are usually rounded (and sometimes the rounding rule is set in the law, as in the Australian case) - in the case of benefit levels in the UK to the nearest 5 pence per week and tax credit levels to the nearest £5 a year. In the case of tax thresholds and allowances, the convention is to round up to the nearest £10 per year except in the case of the basic rate limit and the age allowance income threshold, which are rounded up to the nearest £100 per year. Over time, such rounding can have cumulative effects that create significant changes in values, particularly if the convention involves systematically rounding up or down. For example, over 20 years of uprating by the RPI (as given in Appendix 2), the effect of the rounding convention on the 2006–07 tax allowance would be to increase it by 1.5 per cent more in nominal terms than uprating without rounding would do. In the case of the basic rate income tax threshold, rounding gives rise to a 2.5 per cent higher level over 20 years than from exact uprating alone. However, in the case of Child Benefit, rounding would result in almost no difference in the value after 20 years.

These examples assume a constant inflation rate after the initial period. Fluctuating rates of inflation and growth can create possibly unintended effects. For instance, at the current time, the Basic State Pension is increased in line with changes in the Retail Prices Index, or by 2.5 per cent at a minimum, if inflation is low. Over time, with fluctuating inflation rates, this would be expected to lead to growth in the real value of the pension. Another example is the policy of the 1970s that the pension should increase with the *faster* of prices or earnings. In (unusual) periods when earnings grew more slowly than prices (as in the early 1980s),⁵ this had a 'ratchet' effect, which meant that, over time, the basic pension could have grown faster than earnings.

If inflation and income growth were constant over time, the reference period of the updating index would not be an issue. But in reality they fluctuate over time. This leaves policy-makers with a dilemma. 'Forward-looking' adjustment, applied for example in Hungary and in the UK in the late 1970s and early 1980s, based on *expected* changes, might do better in keeping benefit and tax amounts in line with the intended objective. However, if the forecasts turned out to be wrong, later adjustment might be required (but would be hard to explain). Alternatively, 'backward-looking' adjustment, used in most countries, including the UK today, based on known past changes, would be more robust, but might leave values varying from their intended level – for instance, if inflation sharply accelerated, the purchasing power of those benefits based on past price changes would fall behind.

3 International practice¹

To illustrate the wide range of potential approaches to indexation, this chapter examines current and recent approaches in a number of contrasting countries. Benefit systems in different countries can be extremely diverse in a number of dimensions: the nature and magnitude of different programmes, the tax mix used to finance social support, the relative role of local and central governments in setting the rules and therefore the degree of within-country variation, the incidence of means-testing mechanisms, sensitivity to 'poverty trap' concerns, private pension expansion and so on (Adema, 2006). In countries such as UK, Australia, and Canada, welfare systems are financed mainly through general taxation and characterised by an extensive use of means testing aimed at providing social support only to those deemed in need. Scandinavian countries take a more universal approach, providing support based on citizenship rather than on need. In continental Europe, most programmes are contributory based, therefore eligibility and often the level of entitlements depend on the past contribution history. Also, while in most OECD countries payments levels are defined centrally, as is the case in the UK, there are systems where payments can vary across local areas, as in the USA, Canada, Norway and Switzerland. Moreover, indexation practices may vary for different benefits and single components of benefits within a single country. However, similar types of system are not necessarily characterised by the same types of uprating.

All these reasons make it hard to classify benefit types for comparison purposes (Gutierrez *et al.*, 2005), so that we cannot simply categorise benefits and compare adjustments for each category across countries. The way the benefit is funded, the eligibility restrictions to particular population groups and the base amount to which its level is set could justify different uprating practices, even in the light of the same equity principle (Cantillon and Van Mechelen, 2004).

Adjustment mechanisms across the world

Benefit uprating procedures differ across countries along several dimensions, as summarised in Table 2. Indexation might be required by the law and therefore be applied automatically (as happens in Slovenia, Denmark, Belgium, Australia), or left instead to government or parliamentary discretion (as is the case in Ireland).

Dimensions	Possibilities	Example
Legal requirements	Legal: automatic indexation (timing and formula) Discretionary (government initiative) Combined: Government/Parliament has	Denmark Ireland
	statutory right to uprate	New Zealand
Legal coverage	Overall tax-benefit system Single tax-benefit components	Denmark EU new member states
Role for political process	Statutory provision can be de facto suspended No legal provision but regular de facto adjustment	Belgium Ireland
Adjustment base	Prices – CPI Ad hoc price indices Average wages (net or gross) Social contribution revenues Weighted averages of above measures Ad hoc adjustments (structural reforms)	USA Belgium Netherlands Latvia Czech Republic Estonia
Base measurement timing	Past increase Predicted increase	Australia Hungary
Adjustment timing	Fixed occurrence Trigger mechanisms	Australia Belgium
Extent of differential uprating	Unified system Different rules for different tax-benefit components	Denmark EU new member states

 Table 2 Dimensions of international uprating policy differences

In some cases, the law prescribes both the timing for periodic uprating and the formula to be applied (e.g. under the Australian Social Security Act). Some countries have adopted a combined approach, so that the law requires uprating without specifying the formula or rather gives to the Government the statutory right arbitrarily to modify benefit amounts (as happens for New Zealand benefits, with the exception of pension payments, which are fully regulated). Also, the law might cover the overall benefit system (as in Denmark and Belgium) or just specific parts of it, as happens in most countries. In any case, the existence of a legal requirement for uprating interacts with the political process in such a way that de facto uprating choices might deviate from statutory provisions. On the one side, statutory adjustment might be suspended (as happened in 1999 for Hungarian pensions, or in Belgium in the 1990s) or only partially applied. On the other side, regular adjustments might be applied in countries where there is no statutory provision for automatic indexation, as part of the budget process (in Ireland, Germany and New Zealand, for example). The existence of a legal provision for uprating can, however, contribute to the transparency of political choices, as deviations from the law will need to be discussed explicitly and justified in the public arena.

In addition, different countries adopt various indices as the basis for their adjustments – mostly general price indices, such as the CPI (Canada, USA) or alternative indices excluding particular types of goods (for example, the Belgian 'smoothed health index' excludes tobacco and alcohol); but also average wages (Denmark and Netherlands), social contribution revenues (Estonia, Latvia) or weighted averages of different measures (mostly a combination of prices and wages as for pension payments in Hungary, Czech Republic, Finland, Poland, Slovak Republic, Switzerland). Their development is usually measured over some past period of time, typically the previous year. Occasionally the official predicted increase is used instead (as is the case in Hungary). In the case of wages, the measure can be either gross or net. In practice, net wages are expected to increase more slowly than gross wages on the assumption that, with ageing populations, the tax burden on the working-age group will have to rise (Disney and Johnson, 2001).

The point in time when adjustment occurs is mostly fixed once or twice a year; alternatively, indexation can be triggered automatically when some threshold of change in the index is reached any time during the year (as in Belgium and for the New Zealand pension adjustment).

The extent to which differential uprating is applied varies across countries. Apart from a few examples of unified systems, as in Denmark and Belgium, most countries adjust different components of the system according to different rules. For example, besides a general inflation indexation for benefits, pension payments are adjusted for accounting only, or also for average wage change (e.g. in Australia, New Zealand and Slovenia). Finally, budgetary constraints, standard of living concerns, political acceptance of benefit support and local government discretion play roles to some extent in any country.

Some countries adopt the same indexation rule for all parts of the public pension system (Belgium, Canada, Hungary, Ireland, Norway, USA); other countries use differential uprating for different components (Finland, France, Iceland, Sweden). In some the wage link is used for the targeted scheme (Australia, Iceland, France), according to the principle that real economic growth should be shared partly by those in need, rather than by those who contributed to it. An opposite approach in other countries is to use the price link for the means-tested or minimum pension and the wage link for the earnings-related pension (Sweden), the underlying principle being that only those who contributed to past economic growth should be entitled to share the benefits of current growth.

To clarify how the interaction of all of these dimensions shapes each country's uprating regime we consider three groups of countries as case studies. The first

group covers Denmark, Finland and Sweden, since Scandinavian countries are generally regarded as 'best practice' examples in terms of welfare provision. The second group covers countries such as Canada, Australia and New Zealand whose welfare and political institutions resemble relatively closely those of the UK, allowing us to draw lessons from comparable settings. The third group includes Hungary, Slovenia, Estonia and Latvia, new members in the EU, where new tax-benefit systems and methods of uprating have had to be established recently, and where the relatively high levels of inflation they have experienced have made tax and benefits adjustment a more visible and pressing issue to consider in the design of the overall system. For such reasons, these countries have had the opportunity to adopt 'modern' adjustment practices, unconstrained by rules established in the past but still compatible with public budget requirements.

Nordic countries

Denmark has a unified and comprehensive uprating system where, by law, all benefits and tax parameters are systematically indexed on 1 January to average earnings development (with a lag and small deduction). The indexation is applied to family benefits, invalidity and old age social pensions, employment injuries, unemployment and social assistance benefits.²

In *Finland*, the basic means-tested pension amounts and the parameters of the means test are uprated annually according to prices as expressed by a 'cost of living' index, and the same rule applies to family benefits and to the minimum income non-contributory scheme. In contrast, earnings-related old age pension and employment injuries benefits are uprated by a different index, prescribed by the law to correspond to the weighted average of price and wage changes, where a 20 per cent weight is assigned to wages and an 80 per cent weight to prices (before 2005, a 50–50 per cent mix was in place). Central income tax parameters can be adjusted on a discretionary basis and have been de facto adjusted yearly in recent years.

Similarly to Finland, in *Sweden*, the means-tested guarantee pension amounts are adjusted annually according to prices, while the earnings-related pension payments are uprated according to average earnings growth, less a norm of 1.6 percentage points, in such a way that pension might fall over time in relation to earnings, but grow in real terms. Also, in order to keep the balance between contribution assets and future pension payment liabilities, the uprating index may be reduced to restore balance. Other benefits are either uprated by prices (minimum and maximum amounts for invalidity and employment injuries benefits and sickness and maternity

benefits), or according to a discretionary annual parliamentary decision on an ad hoc basis (minimum and maximum amount for family and unemployment benefits, minimum income, income tax parameters).

Australia, Canada and New Zealand

The 1991 *Australian* Social Security Act generally requires periodic adjustment of all benefits according to the CPI. Although price increases are measured quarterly, payment adjustments are legislated to occur twice or once a year, depending on benefit type. Occasional adjustments to amounts also happen and further provisions can be part of the annual budget process.³ Income tax thresholds are updated in a discretionary way through the annual budget. For example, tax thresholds were unchanged from 2000 to 2002, but subsequently thresholds for the higher income brackets were increased each year from 2003 to 2005. The main departure from the general CPI indexation for benefits is represented by the means-tested old age pension. While statutory provision requires automatic indexation by the CPI, keeping the single person's pension level with at least 25 per cent of male total average weekly earnings (MTAWE) has, since 1999, provided a lower legal limit at the March and September indexation points.

In *New Zealand*, according to the law, benefit amounts and parameters are increased at the discretion of Government, which can also decide the adjustment timing. In practice, amounts of most benefits are uprated annually according to the previous year's increase in CPI, but these increases are claimed politically as budget reforms. In-work benefits are reviewed periodically and adjusted at the discretion of the Government. Another exception is temporary disability benefit, which is increased annually according to a labour cost index. The main exception to the general de facto CPI indexation is the pension, known as the New Zealand Superannuation. According to the law, uprating follows CPI until either a ceiling of 72.5 per cent of average earnings or a 65 per cent floor is triggered. In the first case, with CPI rising more than wages, indexation follows wages when the ceiling is reached; in the opposite case of wages growing faster than prices, again indexation follows wages if the 65 per cent floor is triggered.⁴ Income tax thresholds are not indexed routinely and were not increased between 2000 and 2005.

In *Canada*, adjustment decisions involve both the federal and the provincial governments. Provincial administrations have discretion over the level of social assistance payments, which vary across the ten provinces. External factors such as political acceptance, public finance and concerns over work incentives have

tended to dominate criteria related to standards of living. Non-indexation was used in the 1990s to reduce real benefit payments, bringing the average social assistance payment from 40 per cent of the official poverty line in 1985 to 30 per cent in 2000. The federal government is mainly responsible for pension payments – both the basic and the targeted pensions have been indexed quarterly to the CPI by law since 1973.

New member states of the European Union

In *Hungary*, while different types of pension are uprated regularly according to the law, for the other components of welfare provision there is no statutory requirement for annual adjustment, nor are they regularly adjusted on an ad hoc basis. However, the Hungarian pension offers an interesting example of how general principles, economic contingencies, budget requirements and political acceptance can interact in shaping adjustment practices. From 1991 to 1997, pension uprating was based on expected nominal wages but, when real wages started to increase in 1997, the World Bank suggested the introduction of price indexation. Political concerns led Hungary to opt for an index combining a price and wage adjustment so that pensions would increase more than prices but less than wages. This formula was suspended in favour of projected inflation alone in 1999, but since then Hungary has returned to combined price-earnings indexation, adopting a 50–50 per cent mix.

All benefit uprating mechanisms in *Slovenia* are regulated by law, based mostly on inflation, with some exceptions. An important element of the Slovenian uprating system is known as the 'guarantee wage', used in the means-testing calculations and for the minimum payment in some contributory benefits. Originally, it was the lowest possible pay for a full-time job and currently it amounts to approximately 20 per cent of the average gross wage. It is adjusted each year by 85 per cent of the expected rise in the CPI. The CPI is also used for uprating income tax thresholds, allowances and tax credits. Exceptions to price indexation are found in the case of disability and attendance supplements, calculated as a percentage of the average net wage.⁵ Pensions are uprated twice a year in February and November, according to the average gross wage growth, with 0.5 percentage points subtracted from the rate so that pension growth lags behind wage growth.⁶

There is no general provision in *Estonian* law requiring regular uprating of benefit levels,⁷ but the amounts have been raised recently through discretionary reforms, the main motivation being the phasing in of more generous benefits. While employee social security contributions are calculated as a percentage of gross earnings, the law requires the maximum limit to self-employed social security contributions

to be indexed to the minimum wage. This is set annually after discussion with the Social Partners. A legal act of 2002 requires pensions to be uprated in April of each calendar year by a factor that is the arithmetic average of inflation and the growth of social tax revenue (which in turn depends on economic growth and on the number of people in employment). In recent years, inflation has been much lower than the social tax revenue growth and, while it was argued that this would penalise pensioners through the uprating mechanism, it has also provided politicians with opportunities to award 'extra' pension increases.

Similarly to Estonia, in *Latvia* there is no statutory requirement for uprating family and social assistance benefits; minimum income is adjusted by the Cabinet according to the annual budget. Nevertheless this country offers an interesting example of an uprating formula for old age and invalidity pensions. The annual adjustment depends on the amount of the pension, being more favourable to lower amounts. For amounts lower than three times the state social security benefit, indexation is based on inflation and social insurance earnings (50 per cent of real growth of aggregate wages); prices-only indexation applies if the pension amount exceeds this threshold; and no indexation is applied if the pension amount exceeds five times the state social security benefit. Again, it is difficult to distinguish this regime from an attempt to phase in a more desirable pension structure.

Summary

- There is great variety in uprating practices in the countries we have considered: different countries adopt different approaches; hybrid rules and differential uprating are widespread; and de facto adjustments often replace legal provision and are also used to implement concealed structural reforms.
- Besides a very few examples of automatic and unified earnings uprating, the most common practice remains adjustment linked to prices.
- Most of the uprating policy debate is focused on pension payments.
- When uprating is to some extent linked to earnings, this is limited to pension payments, while the rest of the tax-benefit system parameters are linked to price movements, one way or another.

- Some countries adopt the same indexation rule for all parts of the public pension system; other countries use differential uprating for different components. In some, the wage link is used for the targeted scheme and the prices link for the contributory scheme; in others, the reverse is the case.
- Overall, there does not seem to be a single obvious solution to the choice of most effective uprating approach. Nevertheless, the presence of a legal provision for adjustment appears to contribute to the transparency of political and spending choices, and provides a clear default by which actual practice may be judged.

4 Recent practice in the UK

As already indicated, there is a number of different criteria used in uprating different parts of the UK benefit and tax systems (see Appendix 1 for rules on social security benefits). Since the 1980s, most benefit levels have been uprated annually for inflation, by the Retail Prices Index (RPI), measured looking backwards. Meanstested benefits are uprated by the Rossi index, which excludes housing costs and local taxes on the basis that these costs are supported directly. However, a few – such as the Guarantee Credit for pensioners and the child rates within the Child Tax Credit – are currently adjusted by average earnings.

The legal requirements, as well as actual practice, for uprating also vary both across and within benefit types. There are statutory requirements to uprate some elements annually by prices, while, for other aspects of the same benefit, uprating is discretionary, sometimes leaving parts of the system – such as capital limits and earnings disregards in Income Support – at the same nominal value for years.

The consequences of these differences in uprating practice can be very large, as can be seen in Figure 2, which shows the value of particular benefits as a percentage of average earnings since the 1970s.

Back in the 1970s, a single person received Supplementary Benefit (now Income Support) or flat rate Unemployment Benefit (now Jobseeker's Allowance) worth around 20 per cent of average earnings. By 2004, the equivalent was worth only 11 per cent of average earnings. In the early 1980s, the basic pension was worth a quarter of average earnings. Now its (generally) price-linked value has fallen below 16 per cent of average earnings. Of those benefits shown, only Income Support for younger pensioners (now the Guarantee Credit) has (nearly) regained, and recently held, its relative value of the late 1970s.

Equally, the tax structure has changed over time, as some thresholds have been increased more or less generously than others. Figure 3 shows the points at which different kinds of people would start to pay income tax or make National Insurance contributions, expressed as a percentage of average earnings since 1973. The gradual reduction in value of the thresholds relative to earnings over time is evident. The main single personal income tax allowance has generally been increased in line with price inflation,¹ while the tax thresholds for single pensioners (benefiting from the 'age allowance') have tended to retain their values better in relation to earnings since the early 1990s. In recent years, tax allowances for pensioners have been linked to the value of the Guarantee Credit in order to exempt recipients from income tax and are therefore de facto earnings indexed.



Figure 2 Benefit values as percentage of average earnings, 1971–2006

Source: DWP (2006a).

The former Married Couple's Allowance in income tax was reduced in value even in relation to prices in the 1990s. The married allowance for working-age couples was finally abolished in 2000–01, although many of its recipients were compensated the following year by the introduction of the Children's Tax Credit, which is not shown. This illustrates the difficulty in recording the history of uprating practice, since structural changes to the system ('reforms') occur alongside regular uprating. Another example of this is shown by the increase in the threshold for actually paying National Insurance contributions in 2000–01. This was also an explicit policy change, deliberately bringing the threshold for payments by employees in line with the tax threshold by 2001–02. Indeed, the downward trend in the value of thresholds compared with average earnings should not be taken to imply that tax burdens have risen. Other changes within the income tax structure – notably reductions in tax rates - have more than compensated for fiscal drag, at least on average. For example, the proportion of income taken in income tax of a single childless person on average (mean) earnings (and no other income) fell from 19.7 per cent in 1990-91 to 16.8 per cent in 2001-02, rising again to 17.2 per cent by 2004-05.2

Figure 3 Income tax and National Insurance contribution effective thresholds in relation to average earnings, 1973/74–2005/06



Between 1994–95 and 1999–00 for all married people, and until 2005–06 for those over pension age, calculations show the effective threshold, taking account of the fact that married allowances were/are allowed at a restricted rate.

No account is taken of tax credits or other concessions for children.

Sources: HM Revenue & Customs (HMRC) information on income tax allowances, reliefs and rates, www.hmrc.gov.uk/stats/tax_structure/menu.htm.

Average earnings for full-time adult employees from DWP Annual Abstract Table 3.2, based on New Earnings Survey (NES)/Annual Survey of Hours and Earnings (ASHE).

While most elements of the Child and Working Tax Credits are currently uprated by prices (with some exceptions as discussed below), the history of the uprating of in-work support for parents since the introduction of the Family Income Supplement (FIS) in 1971 cannot be distinguished from the reforms to the structure and generosity of the benefits and credits, and to the conditions under which families may be entitled. For example, the maximum tax credit payment for a lone parent in work for 24 hours a week with two children (both at least 1 year of age) and without qualifying childcare costs in 2005–06 was £137.11 per week, which was 26.0 per cent of average earnings. An equivalent family would have received a maximum of £8 in FIS in 1975–76, which was 13.6 per cent of contemporary average earnings. Clearly, maximum in-work support for this type of family has become much more generous, even in relation to earnings. However, if the family had other characteristics, it might not have been entitled at all in 1975 or the difference in relative generosity at the two points in time might have been different. Here uprating policy for FIS or the tax credits (or Family Credit or Working Families Tax Credit, which came in between) is not the critical issue. In this case, it was the major structural reforms that channelled increasing proportions of the government budget to low-income working families that had the impact.

So, in setting out principles that might underlie the uprating of benefits, credits and tax thresholds, and in drawing out the implications for alternative practices, it is not straightforward to compare actual benefit and tax systems at different points in historical time, or across countries. Rather, we can consider what would have happened or might happen under alternative uprating regimes – including current stated policy – for a particular policy scenario, on the basis that everything else stays the same. This is what we do in Chapters 6 to 9 of this report.

5 Policy in the future, as it looks now

To examine the long-run implications of current indexation conventions, parts of the analysis that follows compare outcomes under the 'current' system (incorporating reforms announced up to the 2007 Budget) with the systems that would emerge if those conventions were continued in the short term (for six years) and in the medium term (for 20 years). In Chapters 6 to 9, *we look at the impacts of the future systems* as if *they had been in force in the base year, 2006–07. In reality, of course, as time goes by, many features of the population and of the income distribution will change, so our results do not represent a forecast of the future, but rather a way of isolating the medium-term implications of different indexation rules, abstracting from such population and other possible changes.* Chapter 10 takes a different perspective. It looks at how current incomes of people over 65 are likely to evolve in practice as they age and also draws comparisons between now, six and 20 years' time in the projected incomes of different age groups within the older population.

Both these approaches require the levels of benefits and other parameters of the tax and benefit system to be projected forward. As Chapter 4 of this report makes clear, there are aspects of current uprating practice that are explicit, with clear government commitments to one approach or another. There are others where reforms have been announced for the future. But, in many cases, actual policy is decided from year to year. For many of these, recent practice has followed implicit rules, which it is reasonable to assume will continue unless an explicit reform is announced. But, for some aspects of the system, it is less clear what 'business as usual' would entail and a judgement has to be made. This chapter summarises the assumptions we have made on which our 'base case' projections in later chapters are based. Details are given in Appendix 2, which also explains the assumptions used in constructing the indices. In summary we have assumed the following.

- *Price indexed (with Retail Prices Index)*: the default assumption for all social security benefit and tax credit amounts and thresholds unless specified below; income tax and National Insurance contribution thresholds.
- *Price indexed (with Rossi index)*: Income Support; Housing Benefit and Council Tax Benefit applicable amounts (except for pensioners and children).
- *Earnings indexed (using an Average Earnings Index)*: child elements of the Child Tax Credit (until 2009–10); Guarantee Credit threshold for pensioners; basic pension (from 2012–13); Savings Credit threshold (from 2009–10 to 2014–15).

• *Fixed in nominal terms*: the family and baby elements of Child Tax Credit; the Child Tax Credit and Working Tax Credit income threshold; capital limits for receipt of means-tested benefits; additional basic pension at 80; winter fuel payments to pensioners; earnings and other disregards in Income Support, Housing Benefit and Council Tax Benefit.

These assumptions take account of the reforms to state pensions announced in the May 2006 White Paper (DWP, 2006b) and enacted in the 2007 Pensions Act, making the assumption that the basic pension is relinked to earnings in 2012 (although this date is not yet a firm government commitment). In extrapolating uprating policies up to 20 years ahead, we have had to make some judgements about how stated policies will apply over time – in particular, the following.

- Income tax age allowance uprating is assumed to be on the same basis as Pension Credit guarantee uprating (i.e. with earnings).
- Elements of benefits that have not been uprated regularly in the past are assumed to be frozen throughout the six and 20 years considered here.
- As implemented in the 2007 Pensions Act, the Pension Credit Savings Credit lower threshold is uprated by earnings from 2008 until 2014. From then on, it is worked out as a function of the maximum payment and the Guarantee Credit level.
- There is a commitment to earnings uprate the Child Tax Credit child amounts until 2009–10. After that time we assume a return to price (RPI) uprating¹. While there is some expectation in policy circles that earnings uprating will be extended for a longer period, there is no stated policy on this, so we retain price uprating as the default.

Aside from these specific assumptions, where there is no stated policy for as long as 20 years ahead, we assume the continuation of whatever is the latest policy.

Further structural reforms to the income tax and National Insurance contribution systems, and to benefits and credits were announced in the 2007 Budget. Our main results incorporate these reforms into both our starting point and the future structures that follow from different indexation regimes. This allows us to isolate the implications of indexation rules as they would unfold within the tax and benefit structures that have already been announced. The main features of this reform are:

abolition of the initial 10 per cent income tax band (in 2008);²

- reduction in the basic rate of income tax from 22 to 20 per cent (in 2008);
- an increase in the Upper Earnings Limit for National Insurance contributions to equal the threshold for payment of income tax at 40 per cent (by 2009), with the latter threshold increased by more than inflation (in 2009);³
- above-inflation increases in values of tax credit thresholds (in 2008) and maximum Child Tax Credit child element (in 2008),⁴ and in the age allowances for income tax (in 2008 and 2011);
- increase in the tax credit taper from 37 to 39 per cent (in 2008);
- a slightly greater increase in Child Benefit for the first child than implied by price indexation alone (by 2010).⁵

More details are given in Appendix 3 and Box 3 in the following chapter provides an analysis of the distributional effects of the structural changes announced in the 2007 Budget. The modelling does not take into account the increases in Child Tax Credit and Child Benefit announced for the first time in the 2008 Budget. Our starting point is illustrated in the case of a hypothetical lone parent in Box 2 at the end of this chapter.

Interpreting the values of components of people's future incomes is complex – for all the reasons that indexation and uprating can be the subject of different principles. One way of presenting results would be in *cash terms*. But the meaning of so many pounds or millions of pounds in 2026 is of limited value – amounts may look very large by comparison with today's incomes, but so will everything else. In the analysis that follows, we present results in three different ways that give more meaningful comparisons with current values.

- We present some results in *real terms*, that is using future cash amounts adjusted back to 2006–07 prices by expected future inflation rates, using the Retail Prices Index. For years from 2009–10, we assume that this inflation rate will be 2.75 per cent annually, but 3.5 per cent in 2007–08 and 2008–09. See Appendix 2, table A2.2.
- Other results are given in *earnings terms*, that is using future cash amounts adjusted back to 2006–07 values by expected future growth in average earnings. For years from 2009–10, we assume that average (cash) earnings will grow by 4.805 per cent annually (i.e. by 2 per cent in real terms). See Appendix 2, table A2.2.

- Some of the projections in the following chapter are shown in relation to *relative poverty lines*. These are assumed to move in line with the growth of median net incomes, as modelled for the base case in Chapter 6. These grow somewhat more slowly than average earnings as a result of the way in which components of median incomes essentially many benefits do so, and as a result of the effects of fiscal drag when tax thresholds increase more slowly than gross incomes.
- Some of the discussion of the impacts on the Government's finances looks at these in terms of changes in the *percentages of national income* (GDP) that taxes and spending would represent.

The main modelling results in Chapters 6 to 9 are essentially in earnings terms. They are calculated by taking the parameters of tax and benefit systems projected under different rules in 2012–13 and 2026–07, adjusted back to 2006–07 values by expected earnings growth, and then applying them to the population distribution of other gross incomes as they actually were in 2006–07. They show what future tax and benefit systems would 'feel like' if applied to current incomes. This allows us to concentrate on the long-run impact of indexation rules, abstracting from other factors that will affect the actual distribution of incomes in the future – such as the ageing population or the maturing of pension rights. It is equivalent to making the assumption that all components of gross income *other than* those set by the State through tax and benefit rules grow in line with earnings growth, and then expressing the results in 2006–07 earnings terms (and abstracting from changes in population composition).

Box 2 The effects of the post-Budget 2007 tax-benefit system on the income of a lone parent with one child

Figure 4 illustrates how the post-Budget 2007 tax and benefit system would affect a hypothetical lone parent with one child if they were working different numbers of hours in 2006–07 at the minimum wage. Their gross earnings would rise from £85 if they worked 16 hours per week to £440 if they worked 80 hours per week at the minimum wage (or equivalently for fewer hours at a higher wage). Working for 30 hours at the minimum wage (£160.50 at 2006–07 earnings levels) corresponds to earnings of around 30 per cent of average full-time earnings). But, as their gross earnings grew, so would their income tax and NIC liabilities, shown in the bottom part of the figure. As a result, their take-home pay, shown in the upper part of the figure, would only grow from £85 to £325. At

the same time, higher net earnings would mean reduced benefits (Council Tax Benefit and Housing Benefit) and reduced tax credits. Allowing for their Child Benefit receipt, but also rent (\pounds 53.92 per week) and council tax (\pounds 14.70), their net income after housing costs (AHC) rises from \pounds 165 per week on 16 hours to \pounds 215 on 40 hours and \pounds 285 per week on 80 hours, shown by the red line. This line corresponds to the 'AHC 2006' line shown in Figure 7 in the next chapter.



Figure 4 Taxes and benefits for a lone parent with a single child by hours of work at minimum wage under the post-2007 Budget system

Source: own calculations, adapted from DWP's (2006c) *Tax Benefit Model Tables* and using DWP assumptions about rent and Council Tax.
6 Implications of current indexation rules for incomes and poverty

In this chapter, we consider the ways in which current indexation rules – the 'base case' described in Chapter 5 – would change the impact of the benefit and tax systems if continued over the short term (over six years to 2012–13) and the medium term (over 20 years to 2026–27). Our starting point is the 'post-reform' system after the structural changes announced in the 2007 Budget, as described in the previous chapter. The effects of the structural changes themselves are set out in Box 3 providing, with a single year's worth of reform announcements, a yardstick against which to compare the effects of base case uprating. We start by showing the impact of base case uprating on a selection of illustrative families. This is followed by analysis of the effects on the population as a whole – on the income distribution and on poverty rates. As explained in the previous chapter, the main analysis is in relative earnings terms and poverty is considered relative to contemporary median incomes. We also consider the implications of base case uprating for poverty measured in absolute terms and for incomes relative to prices (see Box 4 at the end of this chapter).

Box 3 The effects of the 2007 Budget structural change announcements

The 2007 Budget changes (described in Chapter 5) themselves have distributional and budgetary consequences that provide a yardstick against which to compare the effects of uprating, as well as providing the base structures from which we measure change. Using POLIMOD (see Appendix 4), we estimate the budgetary cost of the reforms as £2.11 billion in 2006–07 prices, with households paying £780 million less in income tax and National Insurance contributions (NICs) and receiving £1.26 billion more in tax credits and means-tested benefits. The extra cost of Child Benefit is £75 million.¹ This compares reasonably well with the 2007 Budget Report estimates, which show a cost of £2.43 billion against an indexed base in 2009 (excluding the cost of the Child Benefit increase).²

The distributional impact of the reforms is illustrated in Figure 5, showing the average net gain for each tenth of the income distribution (in order of prereform incomes). On average, each tenth (or decile group) gains, with the largest gains as a percentage of starting income in the second and third tenths of the distribution, amounting to 0.75 per cent of income or more, reflecting the increased generosity of the tax credits. Gains are a smaller percentage of income for higher-income groups, making their overall impact progressive, although the ninth decile group gains more than 0.4 per cent of income, reflecting the gain from the cut in the basic rate of income tax. However, as the discussion above suggests, not all individuals gain from the reforms - overall around a guarter lose from them, as shown in Figure 6.



Figure 5 Distributional impact of the 2007 Budget structural reforms

Decile groups are defined according to the before housing costs household disposable income of individuals before the reforms, equivalised using the modified OECD scale. Source: POLIMOD using Family Resources Survey (FRS) 2003-04.

These are predominantly people in households who either are not entitled to tax credits (for instance childless, working-age households working less than 16 hours per week), who lose more from the ending of the 10 per cent band than they gain from the cut in the basic rate, or who fail to claim the tax credits to which they are entitled.

Table 3 shows the effect of the reforms on the number of people with incomes below poverty lines measured as 60 per cent of median income either before or after housing costs. The first thing to note is that the reforms have a net cost to the Exchequer and so generate a gain to households on average. Median incomes are therefore slightly higher – by 60p per week before housing costs

after the reforms and 40p per week after them – so the post-reform poverty lines are slightly increased. The impact of the reform – if it is implemented all at once – is to cut the overall poverty rate by 0.6–0.7 percentage points, including by around 2 percentage points for children. The pensioner poverty rate is unaffected.





Decile groups are defined according to the before housing costs household disposable income of individuals before the reforms, equivalised using the modified OECD scale. Source: POLIMOD using FRS 2003–04.

Table 3 Relative poverty rates in the UK before and after the 2007 Budgetreforms

	200	6–07	After 2007 B	After 2007 Budget reforms		
%	BHC	AHC	BHC	AHC	BHC	AHC
Median (£ per week)	363.3	310.7	363.9	311.1	0.6	0.4
All	17.2	21.3	16.6	20.6	-0.6	-0.7
Children	19.3	28.8	17.5	26.7	-1.8	-2.1
Pensioners	22.9	16.0	22.9	16.0	0.0	0.0
Working age	15.0	20.1	14.4	19.8	-0.3	-0.3

Poverty rates are calculated as the percentage of people living in households with equivalised income below 60 per cent of the within-scenario median. The modified OECD equivalence scale is used with before housing costs (BHC) incomes and the 'companion' scale is used with after housing costs (AHC) incomes (see DWP, 2007a, p. 189).

Source: POLIMOD using FRS 2003-04.

The implications of future indexation for a lone parent

The effect of six and 20 years of base case uprating of the tax and benefit system on the income of a hypothetical lone parent with one child (see Box 2) is illustrated in Figure 7. This shows what the relationship between hours worked at the minimum wage and the resultant net income after housing costs would look like, measured in relation to a relative poverty line, compared with the starting point of the 'post-reform' system.³





The poverty lines are derived from median incomes reported in Table 4 later in this chapter. Working for 30 hours at the minimum wage (£160.50 in 2006–07 earnings levels) corresponds to earnings of around 30 per cent of average full-time earnings.

Source: own calculations.

Figure 7 shows that net income for any given number of hours in 2012–13 is lower, and lower again in 2026–27. The base case indexation rules mean that this lone parent would over time receive progressively less help from benefits and tax credits, and would pay more tax, measured in relation to contemporary income levels, and hence in relation to the relative poverty line.⁴ While, at the starting point, working

16 hours at the minimum wage would generate a net income 17 per cent above the poverty line, after six years of base case indexation, this would fall to 7 per cent above and, after 20 years, it would be 19 per cent below the poverty line. If current indexation conventions continued for 20 years, this lone parent would need to work for 42 hours per week at the minimum wage for their income to reach the poverty line – compared with a net income nearly 50 per cent above the line for those hours in the starting point. As the figure illustrates, base case indexation implies a system that declines very substantially in its effectiveness in helping to keep a working family of this kind out of poverty.

Implications for other family types

Figures 8, 9 and 10 show incomes in relation to the relative poverty line at the starting point (the post-reform 2006–07 system) and after 20 years of base case indexation for three other family types (equivalent to Figure 7 for the lone parent case). In the initial system, net benefits achieve income for a couple with two children that is close to the poverty line for hours of work between 16 and 48 but, unlike the lone parent, not sufficient to carry them out of poverty. However, after 20 years of base case indexation, they are well below the poverty line unless earnings exceed the equivalent of more than 70 hours per week at the minimum wage. The other two family types – a single person and a couple without dependent children – are further below the poverty line if they have low earnings in the initial case, and even further below after 20 years.

However, Figures 8 to 10 show the interesting phenomenon that, at somewhat higher levels of earnings/hours, net incomes are eventually *higher* in relation to the poverty line (i.e. in relation to median net income) after 20 years than in the initial system. This happens around the point where in-work support is withdrawn in the system at the starting point, for single people corresponding to earnings around 42 per cent of the average full-time earnings and for couples around 55 per cent. Such families at or above those levels of earnings would be paying somewhat more tax in relation to earnings as a result of indexation below the rate of earnings growth and so their net incomes would rise over the 20 years rather more slowly than gross earnings. However, other families would be losing much *more* as a result of benefit erosion, slowing down the growth of *median* net incomes even more. The net incomes of higher-earning childless families who do not qualify for benefits or credits at the starting point therefore rise a little in relation to the relative poverty line.

Figure 8 Income (AHC) for a single person by hours of work at minimum wage: post-reform 2006–07 system, and after six and 20 years of base case uprating (incomes as percentage of relative poverty line)



The poverty lines are derived from median incomes reported in Table 4 later in this chapter. Source: own calculations.

Figure 9 Income (AHC) for a couple with no children by hours of work at minimum wage: post-reform 2006–07 system, and after six and 20 years of base case uprating (incomes as percentage of relative poverty line)



The poverty lines are derived from median incomes reported in Table 4 later in this chapter. Source: own calculations.

Figure 10 Income (AHC) for a couple with two children by hours of work at minimum wage: post-reform 2006–07 system, and after six and 20 years of base case uprating (incomes as percentage of relative poverty line)



The poverty lines are derived from median incomes reported in Table 4 later in this chapter. Source: own calculations.

For those out of work entirely, Figure 11 shows the value of combined Income Support, maximum Child Tax Credit and Child Benefit as a percentage of the (after housing costs) relative poverty line for the same lone parent with one child and the three other family types.

In the post-reform 2006–07 system, these out-of-work benefits would leave the lone parent with an income equivalent to 87 per cent of the poverty line after allowing for housing costs. Under the base case indexation assumptions, this proportion would fall over time so that, by 2026–27, they would only be receiving incomes corresponding to 61 per cent of the poverty line.

For the other family types, out-of-work incomes start further below the poverty line in each case and decline even more rapidly over time, especially for those without children. After 20 years of base case indexation – in this case simply price linking (using the Rossi index) – out-of-work income (Income Support) for a childless couple would be less than a third of the poverty line, compared with just under half at the start. In each case, these show a large deterioration in the ability of the tax and benefit system to protect people from relative poverty. Figure 11 Incomes after housing costs for out-of-work families of different types: post-reform 2006–07 system, and after six and 20 years of base case indexation (incomes as percentage of relative poverty line)



The poverty lines are derived from median incomes reported in Table 4 later in this chapter. Source: own calculations.

Impact on the whole population

The net impact of the falling relative value of benefits and rising tax burdens varies between different family types. The aggregate effect across the whole population therefore depends on the balance of different kinds of family and on the shape of the income distribution. We now examine the implications of the base case indexation assumptions for the population as a whole, driven by the kinds of effect that have been illustrated for particular family types. This analysis uses the tax-benefit model POLIMOD by applying the equivalents of the future systems to the population as modelled for 2006–07 – treating the gradual effects of indexation changes over six or 20 years as if they were implemented as an immediate reform in a single year.

Remember that, in the analysis that follows, it is the *post*-2007 *Budget* system that is used as a base for looking at the long-run implications of indexation conventions. So, for instance, their impact on poverty rates is measured starting from the post-reform base of 16.6 per cent of the population being poor (before housing costs) shown in Box 3.

The distributional effect of current uprating policy compared with what would happen under earnings uprating is displayed in Figure 12, showing the proportionate change in before housing costs income in relative earnings terms for each tenth (or decile group) of the income distribution. The effect of 20 years of current indexation conventions for individuals in the bottom two tenths of the distribution is a fall in income of around a sixth in relative earnings terms. The poorest tenth, for instance, who are largely dependent on cash benefits, see their incomes fall by 17 per cent. The further up the income distribution one looks, however, the more closely net incomes keep up with gross earnings – despite the falling relative value of benefits and, most relevant in this context, the effects of falling relative values of tax thresholds, or fiscal drag. To make another comparison, the gain from the 2007 Budget reforms (illustrated in Box 3) for the poorest parts of the income distribution taken as a whole is barely enough to offset the effects of a single year's worth of erosion of benefit amounts in relative earnings terms.

Continuing to uprate according to current indexation conventions would result in median net incomes falling relative to earnings. As shown in Table 4, median (BHC) income falls by 3.2 per cent over six years and 9.2 per cent over 20 years, relative to earnings. Poverty lines measured in relation to median net incomes therefore also fall slowly in relative earnings terms. However, the implication of the distributional patterns shown in Figure 12 is that the numbers below relative poverty lines would increase. Measured before housing costs, the overall poverty rate would rise from 17 to 19 per cent after six years and to 23 per cent after 20 years, and the trajectory is similar for poverty measured after housing costs. Table 4 shows that the impact on pensioner households is comparatively modest (indeed, after housing costs, the pensioner poverty rate would be slightly lower after 20 years). This reflects the way in which important parts of the benefit and tax system for pensioners are - or will be during the period considered – earnings linked. However, the rise in child poverty is steep and dramatic. On an after housing costs basis, it rises from 27 per cent at the starting point to 31 per cent after six years and 39 per cent after 20 years. The increase in the BHC rate is even more dramatic, rising from 18 per cent at the starting point to 33 per cent after 20 years. Instead of eradicating child poverty, the effect of continuing current uprating policies would be almost to double it.





Decile groups are defined according to the household disposable income of individuals at the starting point, equivalised using the modified OECD scale.

Source: POLIMOD using FRS 2003-04.

Table 4	lelative poverty in the UK after six and 20 years of base case uprating in
relative	arnings terms

	2006–07 starting point		After six years		After 2	After 20 years	
%	BHC	AHC	BHC	AHC	BHC	AHC	
Median (£ per week)	363.9	311.1	352.5	299.6	333.8	279.6	
Poverty rates: all	16.6	20.6	18.9	22.6	23.4	25.4	
Children	17.5	26.7	22.1	31.0	32.8	39.0	
Pensioners	22.9	16.0	24.3	17.3	24.0	14.3	
Working age	14.4	19.8	16.0	21.1	19.6	23.7	
Poverty gap (median) %	18.6	22.9	17.8	25.4	22.1	35.3	

Poverty rates are calculated as the percentage of people living in households with equivalised income below 60 per cent of the within-scenario median. The modified OECD equivalence scale is used with before housing costs (BHC) incomes and the 'companion' scale is used with after housing costs (AHC) incomes (see DWP, 2007a, p. 189).

Source: POLIMOD using FRS 2003–04.

The median 'poverty gap' as a measure of the depth of poverty shows that those who are poor after 20 years of base case uprating are further below the poverty line than the typical person in poverty is at the start of the process.⁵ After housing costs, 20 years of current indexation would mean the typical poverty gap rising from 23 to 35 per cent of the poverty line.⁶ Not only would the numbers in relative poverty have risen substantially, but so would the depth of their poverty.

Again, to compare with the impact of the 2007 Budget reforms illustrated in Box 3, those reforms would be enough to offset only about two years' worth of the increase in relative poverty implied by current indexation conventions.

As explained above, these are not *predictions* of what will happen to child poverty by 2011–12 or 2026–27. They are estimates of the implications of continuing to uprate according to current policy and conventions, other things being equal. Other studies have attempted to predict the effects of changing demographic and employment factors (among others) on child poverty. For example, Brewer et al., (2006a) suggest that in the most favourable case demographic and employment changes could almost compensate for the adverse effects of uprating policy, while Buck et al., (2007, Table 11) estimate much smaller compensating effects. By its nature, predicting changes in population characteristics that are important to child poverty, such as household composition or parental employment, is not an exact science. While it is clear that at least some of the adverse effects of base case uprating on child poverty could be mitigated by other changes, this does not mean that the implications of base case uprating should be of lesser concern. Possible beneficial changes in the population and in employment rates provide an opportunity to reduce child poverty rather than a justification for complacency about the situation possibly not getting much worse.

Considering the effects of uprating relative to income growth, as we have done above, is consistent with the current policy commitment to reduce poverty in relative terms (for families with children and pensioners, at least). This involves comparing the evolution of incomes after taxes and benefits with what happens to incomes at the median. However, an alternative approach, which is consistent with a focus on poverty measured against a fixed threshold, or 'absolute' poverty⁷ is to consider changes in relative prices (or 'real') terms rather than relative earnings terms. The implications of base case uprating for this approach are drawn out in Box 4 at the end of this chapter.

Budgetary effects

Corresponding to households being worse off on average relative to earnings under base case uprating, there would be a gain to the public finances in relative terms. Indeed, by comparison with the starting point of the post-reform 2006–07 tax-benefit system, continuing with prevailing uprating policy for six and 20 years would generate substantial benefits to the government budget compared with comprehensive uprating in line with earnings. Overall, the budgetary effect is a government saving in 2006–07 terms of £17 billion after six years and £47.4 billion after 20 years, or around 1.3 per cent of GDP after six years and 3.6 per cent after 20 years.⁸ This gives a measure of the very substantial extent to which eroding benefit levels and rising tax burdens would ease the pressures on the government budget, other things being equal. Other factors – such as the effect of an ageing population on the demands on the NHS – might, of course, be expected to push in the other direction at the same time.

Box 4 Impact in real terms of current indexation conventions

Often, when comparing the effects of different tax-benefit systems over time, it is assumed that the counterfactual would be to uprate all elements of the taxbenefit system by prices, and changes are therefore considered in real terms. Figure 13 shows the distributional impact in real terms of six and 20 years of uprating using the base case uprating described in Chapter 5, and applied from the starting point of the post-reform 2006–07 system. In these terms, each decile group gains on average, although the size of the gain is very small in the upper decile groups (e.g. after 20 years, £2.68 per week on average in the top decile group). The gain in income in real terms for the bottom income group averages 16 per cent by the end of the 20 years.

The proportions of people gaining and losing from 20 years of base case indexation when measured in this way are shown in Figure 14. It is perhaps striking that there are substantial numbers of losers even in real terms. In particular, in the lower half of the income distribution, around 30 per cent of individuals gain in these terms, but more – over half in the second and third decile groups – lose. The gainers are those affected by factors such as the earnings linking of the Guarantee Credit or the child element of Child Tax Credit. The losers are those affected by elements of the system that are frozen in real terms, or that increase less rapidly than the RPI, for instance because they are linked to the Rossi index, which excludes housing costs.



Decile groups are defined according to the before housing costs household disposable income of individuals at the starting point, equivalised using the modified OECD scale.

Source: POLIMOD using FRS 2003–04.





Decile groups are defined according to the before housing costs household disposable income of individuals at the starting point, equivalised using the modified OECD scale. Source: POLIMOD using FRS 2003–04.

The effects in real terms on the income of our illustrative lone parent are shown in Figure 15. At high numbers of working hours, their net real income would be higher in 2012–13 than at the starting point and higher again in 2026–27. This is welcome, but hardly surprising given that real hourly gross earnings, reflected in the lone parent's minimum wages, would be rising by nearly 50 per cent over the 20 years. However, for earnings up to the equivalent of 50 hours of work at the minimum wage, they would actually be worse off in real terms after the accumulated effects of uprating over 20 years. This is because the tax credits start to be tapered away at lower levels of real income as the threshold is not indexed at all.⁹ The income of this lone parent remains clear of the absolute poverty line, but the margin against even this unambitious standard is squeezed significantly after 20 years.

Figure 15 Income (AHC) for a lone parent on the minimum wage by hours of work: post-reform 2006-07 system, and after six and 20 years of base case uprating (incomes as percentage of absolute poverty line)



In parallel with this analysis of income changes in real terms, we can also look at changes in poverty rates measured against a constant real poverty standard (i.e. against the kind of 'absolute' poverty line used by the US Government, as opposed to the relative poverty line more generally used in Europe). In this case, two factors push down the poverty rate: real increases in earnings and other components of original income; and changes in the tax and social security systems that are more generous than price indexation. Table 5 shows the significant reductions in numbers below poverty lines measured in this way. Overall, the absolute poverty rate (before housing costs) would fall from 17 per cent in the post-reform 2006–07 system to 13 per cent after six years and only 7 per cent after 20 years. For children, the fall is from 18 to 7 per cent and, for pensioners, from 23 to just 4 per cent. If we thought of 'poverty' only in such absolute terms, growing real earnings and indexation of just some elements of the system more generously than with prices is enough to achieve guite substantial cuts in poverty. However, neither current government policy (recently endorsed by the main opposition party) in terms of child poverty objectives, nor public opinion measured in various ways (Hills, 2004, section 3.5) currently takes such an austere view of the objective.

	2006–07 sta	2006–07 starting point		After six years		After 20 years	
%	BHC	AHC	BHC	AHC	BHC	AHC	
All	16.6	20.6	13.1	17.4	7.4	13.7	
Children	17.5	26.7	13.0	23.5	8.8	19.7	
Pensioners	22.9	16.0	17.4	9.3	4.3	3.5	
Working age	14.4	19.8	11.8	17.6	7.8	14.6	

Table 5 Poverty rates after six and 20 years of base case upratingmeasured against constant real poverty lines

See Table 3 in Box 3. Poverty lines after six and 20 years are calculated by indexing the starting point poverty lines by the assumed change in the RPI. Source: POLIMOD using FRS 2003–04.

Summary

While we have been discussing this in dispassionate technical terms, the main conclusion of this chapter is the strikingly large impact of such changes – in particular, the following.

- Ad hoc changes can have significant impacts. For instance, the 2007 Budget reforms imply income gains of half a per cent or more for those at the bottom of the income distribution and, other things being equal, would cut overall poverty rates by 0.6–0.7 percentage points and the child poverty rate by around 2 percentage points.
- However, the accumulated effects of indexation decisions can be far larger. Reforms of the scale of those announced in the 2007 Budget are barely enough to offset the effects of a *single year's* worth of erosion of the value of benefits on the relative income of the poorest income groups.
- Accumulated over a number of years the effects magnify. For the poorest tenth of the population, who are largely dependent on cash benefits, incomes would fall by an average of 17 per cent in relative earnings terms after 20 years.
- Measured on a before housing costs basis, the child poverty rate would rise from 18 per cent at the starting point to 33 per cent after 20 years of base case uprating. Instead of eradicating child poverty, the effect of continuing current uprating policies would be almost to double it. Structural reforms of the scale of the 2007 Budget changes would be required every two to three years just to offset this rise in child poverty.
- After housing costs, 20 years of current indexation of both benefits and taxes would mean the typical poverty gap rising from 23 to 35 per cent of the poverty line. Not only would the *numbers* in relative poverty have risen substantially, but so would the *depth* of their poverty.
- The further up the income distribution one looks, however, the more closely
 net incomes keep up with gross earnings despite the falling relative value of
 benefits and, most relevant in this context, the effects of fiscal drag. A single
 childless person earning at least 42 per cent of full-time average earnings would
 be further above the poverty line after 20 years of base case uprating than at the
 starting point.
- At lower levels of income, and for families qualifying for benefits or tax credits at the starting point, incomes fall relative to the poverty line. For example, a couple with two children and one parent working for 40 hours on the minimum wage would see their income fall from 96 to 67 per cent of the poverty line after 20 years of base case uprating.

 Overall, the budgetary effect is a substantial gain to the public finances, compared to what would happen if the system was adjusted in line with earnings growth (which is what, other things being equal, would mean unchanged revenue and spending in relation to national income). In 2006–07 earnings terms, this amounts to £17 billion after six years and £47.4 billion after 20 years, or around 1.3 per cent of GDP after six years and 3.6 per cent after 20 years.

7 Implications of current indexation rules: work incentives and reliance on means-tested benefits

As a corollary of making the future benefit systems relatively less generous to those with low incomes, we might expect that aspects of work incentives inherent in the system would improve. We consider two aspects of work incentives, the incentive to work at all and the incentive to work more, for those already in work.

First, the incentive to work at all will be improved if the income earnings and in-work benefit/tax credit package of the people concerned is growing faster than out-of-work benefit incomes. Figure 16 shows the ratio of (AHC) household income out of work to that in work (termed the replacement rate) for the four illustrative families considered in the previous chapter. In each case, it is assumed there is one person in work for 30 hours per week and that they are paid the minimum wage.

Figure 16 Replacement rates for illustrative family types at the 2006–07 starting point and after 20 years of base case uprating



The poverty lines are derived from median incomes reported in Table 4 in Chapter 6. Source: own calculations based on in-work earnings corresponding to 30 hours at the National Minimum Wage. The comparison shows the replacement rate at the starting point and after 20 years of base case uprating. In every case, the illustrative families are better off with a person in work – the replacement rate is less than 100 per cent. The extent to which this is so increases slightly after 20 years of base case uprating for the single childless person (the replacement rate falls from 52 to 50 per cent) and it remains unchanged for the couple with two children. However, it rises a little for the childless couple (from 66 to 69 per cent) and quite substantially for the lone parent (from 64 per cent at the starting point to 72 per cent after 20 years). The reason for this is that in-work benefits and credits are falling in value relative to earnings. Out-of-work benefits are falling in value too, but the freezing of the tax credit threshold, as well as earnings disregards and capital limits, means that in some cases in-work incomes fall faster. Unless this issue is addressed, such measures of work incentives suggest that, other things being equal, lone parents would be *less* likely to take minimum wage paid work after 20 years of base case uprating than they are currently.

Second, the erosion of means-tested benefits and credit thresholds and amounts, along with the lowering of tax thresholds, will have an effect on the extent to which marginal increases in earned income are reduced through taxation and withdrawal of benefits or credits. Marginal effective tax rates (METRs), or the proportion of any additional earnings lost in this way, indicate the incentive to work a little more. Table 6 shows the distribution of METRs on an additional hour worked per week at the starting point, and then after six and 20 years of base case uprating. On the one hand, the effect of 20 years of base case uprating is to increase METRs on average by 1 percentage point (from 32 to 33 per cent) and by a substantial amount (5 percentage points or more) for a much larger group (23 per cent) than those who see their METR fall by at least the same amount (9 per cent). However, this is not driven by increases in the numbers with very high METRs because of benefit withdrawal. The proportions on METRs above 50 per cent fall from 9.5 to 6.5 per cent, and it is falling relative tax thresholds that have the main effect. The proportions of people facing low rates (below 20 per cent) fall from 17 to 13 per cent after 20 years, as more people cross the threshold to pay basic rate tax. The proportion of people in paid work facing marginal rates of 40 to 50 per cent, indicating in most cases a move into paying higher rate tax at the margin, more than doubles from 11 to 24 per cent.

This is reflected in the change in the marginal rate faced by people in work at different points in the household income distribution, shown in Figure 17. At the starting point, this is highest for workers in the bottom decile group (49 per cent on average). After 20 years of base case uprating, the disincentive to work more, as indicated by the METR, remains highest in the bottom decile at 39 per cent on average, but nevertheless falls considerably. The reduction in the average METR is due to fewer people facing combined high rates of benefit withdrawal. In the middle of the distribution, the effects of less benefit withdrawal are mitigated to some extent

by the effects of fiscal drag, the combination resulting in small reductions in METRs on average. At the top of the distribution, the average METR is higher than under the starting point, with the effect of the falling relative value of the tax threshold outweighing the effect of the falling upper limit on National Insurance contributions.

		Base cas	e uprating
Range of METR (%)	Starting point	After six years	After 20 years
0 to 19.99	16.7	15.4	12.7
20 to 29.99	30.7	29.3	25.4
30 to 39.99	31.9	32.6	31.0
40 to 49.99	11.2	14.4	24.4
50 to 69.99	5.5	4.6	3.2
70 to 89.99	3.3	3.0	2.5
90+	0.7	0.7	0.8
All	100.0	100.0	100.0
Mean METR %	31.8	32.0	32.7
Median METR %	31.0	31.0	31.0

Table 6 Distribution of individual marginal effective tax rates (METRs) under
the 2006–07 starting point, and after six and 20 years of base case uprating:
percentage of people in work

Source: POLIMOD using FRS 2003-04.

Marginal effective tax rates are calculated for each individual in paid work for a one hour per week increase in gross earnings, taking account of all effects on household disposable income.

Figure 17 Marginal effective tax rates for individuals in paid work by household income decile group: at the 2006–07 starting point and after 20 years of base case uprating



Marginal effective tax rates are calculated for each individual in paid work for a one hour per week increase in gross earnings, taking account of all effects on household disposable income. Decile groups are defined according to the household disposable income of individuals at the starting point, equivalised using the modified OECD scale.

Source: POLIMOD.

We also consider the associated issue of the extent of dependence on means-tested benefits. On the one hand, means testing is seen as problematic over and above its contribution to high METRs through benefit withdrawal as income rises. Not only may recipients feel stigmatised but also, particularly in the case of pensioners, there is a view – reflected in the provisions of the Pensions Act (DWP, 2006b) – that high levels of dependency on means-tested payments reduce the incentive to provide for one's own support, in this case in retirement. On the other hand, if the alternative to receipt of such benefits is simply lower income, then rates of receipt can also be indicative of the extent to which income is underwritten by the means-tested system. Thus receipt of a means-tested benefit or credit is viewed here neither as purely a positive nor a negative feature, simply as an indication of how the system as a whole is working.

Over all types of means-tested benefit and tax credit,¹ the proportion of people in receipt falls from 43 per cent at the starting point to 30 per cent after 20 years of base case uprating (Table 7). This reduction is not because there are fewer pensioners in receipt of Pension Credit (PC). The proportion remains roughly constant, reflecting the indexation of the Guarantee Credit within PC by earnings over the whole period.² Nor is it due to reductions in the number on Income Support (IS), shown in Table 7 in terms of the number of children affected. While base case uprating reduces the relative size of IS payments, this has only a small effect on the number of recipients, as most working-age IS recipients do not have substantial other incomes to fall back on. The main change is due to the numbers on in-work means-tested supplements being dramatically reduced as the size of payments on average, as well as the income threshold for tapering payments, falls relative to family income. The proportion of children in families receiving tax credits would be reduced to less than half within 20 years of base case uprating.

Table 7 Proportion of people in families in receipt of means-tested benefits or tax credits at the 2006–07 starting point, and after six and 20 years of base case uprating (%)

	Base case uprating				
	Starting point	After six years	After 20 years		
All on any benefit or credit*	43	39	30		
Children on CTC with IS	17	16	16		
Children on CTC alone or with WTC**	49	40	23		
Pensioners on PC***	32	34	33		
Pensioners on any benefit	45	46	45		
All on HB/CTB	21	21	21		

* Income Support (IS), Housing Benefit (HB), Council Tax Benefit (CTB), Pension Credit (PC), Child Tax Credit (CTC), Working Tax Credit (WTC).

** Not including cases receiving the CTC family element only.

*** Either Guarantee Credit or Savings Credit, or both.

Source: POLIMOD using FRS 2003-04.

It should be noted that the proportions on Housing Benefit and Council Tax Benefit do not fall at all. For pensioners, entitlement to these benefits is aligned with Pension Credit. For others, while income thresholds may be falling in relative earnings terms, we assume in our modelling that maximum amounts – corresponding to rent and Council Tax respectively – remain buoyant with earnings. Although some beneficiaries would find their entitlements dwindling as their incomes rose relative to the thresholds, which are kept constant in real terms only, others would find these benefits filled the gap left by Child and Working Tax Credits, the thresholds of which would be falling in value relative to prices.

Summary

- To the extent that base case uprating represents a reduction in the relative value of benefits, we might expect improvements in work incentives after the effects have accumulated as measured by effective marginal tax rates and replacement rates. This is only partly the case.
- Incentives to take work at all do not necessarily improve because, while out-ofwork benefits fall in relative earnings terms, in-work benefits fall faster due to the freezing of the tax credit thresholds. For example, a lone parent would have an income out of work that is 64 per cent of that while working for 30 hours on the minimum wage at the starting point, but this proportion – or 'replacement rate' – rises to 72 per cent after 20 years.
- Incentives to earn more improve at the bottom of the income distribution and the number on high METRs (50 per cent or more) due to benefit withdrawal would fall from 9.5 per cent to 6.5 per cent of the working population.
- For the rest of the population, the erosion of tax thresholds has the major effect. In particular, the proportion of people in work paying income tax at the highest marginal rate (40 per cent) rises from 11 to 24 per cent. Marginal rates rise overall in the top income groups and a much larger group of the working population (23 per cent) would find their marginal rates rising noticeably (by more than 5 percentage points) than see them fall by this much (9 per cent).
- The proportion of the population reliant on any means-tested benefit (or tax credit) falls from 43 per cent to 30 per cent after 20 years. Much of this effect is because of reductions in the numbers receiving tax credits. While this may contribute to improving the incentives of some to work more, this is at the cost of increasing child poverty rates, as described in the previous chapter.

8 Implications of current indexation rules: fiscal drag and benefit erosion

Part of the effects described above result from failure to uprate some benefits and tax credits in line with earnings or median income growth, part is due to the indexation of most income tax and NIC thresholds to prices rather than earnings. We have termed the effect of the indexation regime leading to rising tax burdens 'fiscal drag' and the corresponding reduction in the relative value of benefits 'benefit erosion'. In this chapter, we consider the separate contribution of each. We also establish the extent to which the effects of the current indexation regime are driven by failure to index some elements at all (rather than being indexed to prices) and examine how much of a difference is made by indexation of some elements by earnings over some or all of the periods considered.

The contributions of fiscal drag and benefit erosion

Twenty years of current indexation conventions for tax thresholds and allowances (fiscal drag) as applied to the 2006–07 income distribution, while assuming that benefits and tax credits remain constant in relation to earnings, would have the effect of raising £20.2 billion per year after 20 years (measured in earnings terms). This is two-fifths of the government revenue gain from the combined effects. Not surprisingly, this bears much more heavily on higher- than lower-income households, as shown in Figure 18. Indeed, nearly all of the negative effects in the top decile groups are due to fiscal drag rather than the loss in the relative value of benefits.

We can also consider the mirror image situation where tax and contribution thresholds keep pace with earnings but benefits and tax credits remain uprated as in the base case. The distributional effects of this 'benefit erosion' are also shown in Figure 18. Here it is the lower-income households that are most affected – the 16 per cent losses due to benefit erosion shown here for the bottom two tenths accounting for nearly all of the losses from the combined change.

Not only does benefit erosion play a much bigger role than fiscal drag at the bottom of the distribution (and the reverse at the top), but also the aggregate size of the benefit erosion effect is larger. While fiscal drag would be raising £20.2 billion in additional government revenue per year by 2026–27 relative to earnings indexation, benefit erosion would be reducing spending by £27.2 billion per year, or 57 per cent

of the total gain to the public finances. Combined, they have a proportional effect on incomes that is four times the size for the bottom 20 per cent of the population as it is for the top 20 per cent. Moreover, it is clear that reducing or eradicating fiscal drag would do very little to lessen the effect at the bottom. Addressing benefit erosion, one way or another, is necessary if this effect is to be avoided.





Decile groups are defined according to the household disposable income of individuals at the starting point, equivalised using the modified OECD scale.

Source: POLIMOD using FRS 2003-04.

Contributions to poverty rates and poverty gaps

These distributional effects have implications for poverty measures, as shown in Table 8. Looking at the two effects separately, fiscal drag for 20 years causes relative poverty rates to be slightly *lower* than they would be otherwise (if all benefits, credits and tax and NIC thresholds were indexed with earnings). It has little effect on the incomes of the poor or those on the margins of poverty, but reduces median incomes, and hence depresses the relative poverty line slightly compared to the base case. By contrast, if there were no fiscal drag, but benefits and tax credits were uprated under current conventions, relative poverty rates would be slightly higher than from the two effects combined. As Figure 18 suggests, it is benefit erosion that is most important in thinking about the impact of policy choices on poverty rates. Without any benefit erosion (but with tax thresholds continuing to be uprated as in the base case), poverty rates would actually be somewhat lower than they are at the starting point.

	Starting point	Base case	Fiscal drag only	Benefit erosion only
BHC median (£)	363.9	333.8	353.1	344.1
Poverty rates: all	16.6	23.4	15.7	23.9
Children	17.5	32.8	15.9	33.1
Pensioners	22.9	24.0	21.5	26.1
Working age	14.4	19.6	13.9	19.8
Poverty gap (median) (%)	18.6	22.1	17.9	22.9
AHC median (£)	311.1	279.6	299.7	291.2
Poverty rates: all	20.6	25.4	19.6	26.4
Children	26.7	39.0	25.2	39.1
Pensioners	16.0	14.3	13.6	19.1
Working age	19.8	23.7	19.3	23.9
Poverty <i>gap</i> (median) (%)	22.9	35.3	23.4	35.2

Table 8 Relative poverty in the UK after 20 years of base case uprating	y, and under
fiscal drag and benefit erosion alone	

Poverty rates are calculated as the percentage of people living in households with equivalised income below 60 per cent of the within-scenario median. The modified OECD equivalence scale is used with before housing costs (BHC) incomes and the 'companion' scale is used with after housing costs (AHC) incomes (see DWP, 2007a, p. 189).

Source: POLIMOD using FRS 2003-04.

Contributions to changes in marginal tax rates

The view that benefit erosion needs to be tackled if its unequal effects are to be avoided might, from some perspectives, be tempered if benefit erosion were to have dramatically positive effects on work incentives. The effect of the two components on marginal effective tax rates, and hence incentives to work more intensively, is clearly shown in Table 9. Fiscal drag has the effect of increasing marginal rates on average. The mean would rise from 32 per cent at the starting point to 35 per cent after 20 years. Benefit erosion does have the opposite effect. As entitlements fall relative to income, fewer people are entitled and face high effective marginal rates due to withdrawal of benefit. Compared with the starting point, the average marginal rate would be about 2 percentage points lower after 20 years. The proportion of the workforce with marginal rates above 50 per cent would fall from 9.5 to 5.8 per cent. This is a significant improvement, but it is questionable whether it could bring about increases in work effort that would compensate for the loss in income through benefit erosion and, if not, whether it could justify the huge increase in poverty rates.

		Base case uprating after 20 years				
Range of METR (%)	Starting point	Combined	Fiscal drag	Benefit erosion		
0 to 19.99	16.7	12.7	11.8	17.7		
20 to 29.99	30.7	25.4	23.9	32.5		
30 to 39.99	31.9	31.0	29.2	33.4		
40 to 49.99	11.2	24.4	24.8	10.6		
50 to 69.99	5.5	3.2	5.7	3.0		
70 to 89.99	3.3	2.5	3.8	2.3		
90+	0.7	0.8	0.8	0.5		
All	100.0	100.0	100.0	100.0		
Mean METR (%)	31.8	32.7	34.7	29.7		
Median METR (%)	31.0	31.0	31.0	29.5		

Table 9 Distribution of individual marginal effective tax rates (METRs) after 20 years of base case uprating, and under fiscal drag and benefit erosion alone: percentage of people in work

Marginal effective tax rates are calculated for each individual in paid work for a one hour per week increase in gross earnings, taking account of all effects on household disposable income. Source: POLIMOD using FRS 2003–04.

The effects of 'pure' price indexation

Some of the potentially adverse effects of the current uprating regime on poverty, as well as some of the implications for marginal tax rates and the numbers affected by means testing, may be due to the fact that some elements of the system are frozen in value in nominal terms. It is of interest to know how much of the effects of fiscal drag or benefit erosion identified above would be reduced if these elements were uprated using the RPI. We term this 'minimum price indexation'.

At the same time, some elements *are* uprated by earnings and it is equally of interest to know the extent to which this prevents the picture looking worse. We explore this by modelling the uprating of all elements (including those currently frozen) by the RPI.¹ We term this 'comprehensive price indexation'.

Minimum price indexation raises 6 per cent *less* revenue than the base case after 20 years, while the net effect of the comprehensive price uprating scenario is to raise 12 per cent more net revenue than the base case. The distributional effect of these two scenarios after 20 years is shown in Figure 19. Minimum indexation by prices has the effect of reducing the loss relative to earnings, but comprehensive price indexation increases the loss, particularly in the lower decile groups where beneficiaries of the Pension Credit are located.



Figure 19 Relative earnings effects of price indexation after 20 years

Decile groups are defined according to the household disposable income of individuals at the starting point, equivalised using the modified OECD scale.

Source: POLIMOD using FRS 2003–04.

Not surprisingly, pensioner poverty rates would rise dramatically after 20 years of comprehensive price indexation, compared with the base case scenario (Table 10). On an after housing costs basis, the pensioner poverty rate would almost double to 29 per cent. Comprehensive price indexation has a marked effect on lowering the median. So the poverty rates of all other groups are lower by 1 or 2 percentage points than under the base case. Minimum price indexation has a modest effect, lowering all poverty rates a little, and by 1 percentage point for children and pensioners on an after housing costs basis.

Allowing all elements of the system to be indexed by prices at least is of particular relevance when considering the effect of base case uprating on Child Tax Credit (CTC) and Working Tax Credit (WTC). The income threshold at which these credits start to be tapered away is frozen in our analysis (as in current policy) and one plausible explanation for the very sharp increases in child poverty implied by base case uprating (shown in Table 4 in Chapter 6) might be that it is this, rather than the indexation of the CTC child amount only to prices from 2009, that so reduces

the effectiveness of CTC. In fact, this is not so. Amending the 20-year base case assumptions to uprate the CTC/WTC threshold by the RPI (but leaving other frozen elements alone) would lower the AHC child poverty rate by less than 1 percentage point (0.7) and the BHC child poverty rate by just 1 percentage point, from 33 to 32 per cent (compared with 18 per cent at the starting point).²

	Starting point	Base case	Minimum price indexation	Comprehensive price indexation
BHC median (£)	363.9	333.8	335.5	329.1
Poverty rates: all	16.6	23.4	22.7	23.8
Children	17.5	32.8	31.4	30.8
Pensioners	22.9	24.0	23.4	30.2
Working age	14.4	19.6	19.2	19.2
Poverty gap (median) (%)	18.6	22.1	22.4	25.1
AHC median (£)	311.1	279.6	281.9	275.5
Poverty rates: all	20.6	25.4	24.8	27.8
Children	26.7	39.0	38.1	37.6
Pensioners	16.0	14.3	13.1	28.5
Working age	19.8	23.7	23.5	23.8
Poverty gap (median) (%)	22.9	35.3	34.5	34.8

Table 10 Effects of price indexation on poverty after 20 years in relative earnings terms, compared with the base case

'Minimum price indexation' unfreezes all frozen elements of the system and indices by RPI.

'Comprehensive price indexation' indexes all elements by RPI (or Rossi if currently used).

Poverty rates are calculated as the percentage of people living in households with equivalised income below 60 per cent of the within-scenario median. The modified OECD equivalence scale is used with before housing costs (BHC) incomes and the 'companion' scale is used with after housing costs (AHC) incomes (see DWP, 2007a, p. 189).

Source: POLIMOD using FRS 2003-04.

The small effects on poverty rates of price indexation of all elements are a reflection of the small increase in numbers in receipt of some means-tested benefits, relative to the base case, shown in Table 11. Minimum price indexation increases the proportion of pensioners in receipt of any benefit, for example, because this reduces the limiting effects of capital thresholds in Housing Benefit and Council Tax Benefit that are frozen under the base case scenario. Table 11 also shows how comprehensive price indexation greatly lowers the proportion of pensioners in receipt of Pension Credit (to 10 per cent compared with one-third after 20 years under base case uprating). However, the proportion of pensioners on any benefit does not fall so much because Housing Benefit and Council Tax Benefit retain a greater role in the same manner as they do for working-age people as the value of the tax credits erodes under base case uprating (Table 7 in Chapter 7).

	Starting point	Base case	Minimum price indexation	Comprehensive price indexation
All on any benefit or credit	* 43	30	31	28
Children on CTC with IS	17	16	16	16
Children on CTC alone or	with			
WTC**	49	23	22	22
Pensioners on PC***	32	33	33	10
Pensioners on any benefit	45	45	47	33
All on HB/CTB	21	21	22	21

Table 11 Pro	portion of people ir	n families in rece	ipt of means-teste	ed benefits or
credits after 2	20 years of base ca	se uprating and	price indexation (%)

* Income Support (IS), Housing Benefit (HB), Council Tax Benefit (CTB), Pension Credit (PC), Child Tax Credit (CTC), Working Tax Credit (WTC).

** Not including cases receiving the CTC family element only.

*** Either Guarantee Credit or Savings Credit, or both.

Source: POLIMOD using FRS 2003-04.

Summary

- The size of benefit erosion in terms of spending foregone is larger than the size of fiscal drag in terms of extra revenue. The ratio is 57:43 after 20 years of base case uprating.
- Benefit erosion plays a much bigger role than fiscal drag at the bottom of the distribution (and the reverse at the top).
- Combined, they have proportional negative effect on incomes that is nearly four times larger for the poor than for the rich. The 20 per cent of households with the lowest incomes on average lose 17.3 per cent of the incomes relative to earnings. The 20 per cent of households with the highest incomes lose 4.8 per cent.
- Reducing or eradicating fiscal drag would do nothing to lower poverty rates relative to the base case. It is benefit erosion that is most important in thinking about the impact of policy choices on poverty rates.
- Fiscal drag has the effect of increasing marginal deduction rates on average. Benefit erosion has the opposite effect. As entitlements fall relative to income, fewer people are entitled to or face high effective marginal rates due to withdrawal of benefit. The proportion of the workforce with marginal rates above 50 per cent would fall from 9.5 to 5.8 per cent after 20 years of benefit erosion.

- It is unlikely that this improvement alone could bring about increases in work effort that would compensate for the loss in income through benefit erosion. It is therefore questionable whether it could justify the huge increase in poverty rates.
- If all thresholds and benefits were uprated uniformly by prices, the main effect would be a dramatic rise in the pensioner poverty rate, compared with the base case scenario, in which the Guarantee Credit and (from 2012) the Basic Pension are earnings linked. On an AHC basis, it would almost double from 16 per cent at the starting point to 29 per cent after 20 years.

9 Implications of alternative indexation strategies

The analysis in the preceding two chapters does not represent a forecast of what policies will actually be implemented over the next six or 20 years. Some of the substantial revenue gains that current indexation conventions would generate might be used to keep the government accounts in balance without tax rises in the face of other pressures, such as improving public finances to meet the 'Golden Rule', or the pressures from an ageing population. However, parts at least of them may in the event be 'given back' through periodic tax and benefit reforms or ad hoc tax 'cuts'. This has certainly been the experience of the last ten years. What the analysis does do, however, is to indicate the scale of such reforms or adjustments that might be needed even just to prevent poverty rates from rising, let alone to make progress in reducing poverty.

This chapter explores the distributional implications of what would happen if the revenue gains were offset through other changes to the tax or benefit systems. To do this, we consider a series of illustrative scenarios, as follows:

- a. uprating benefits and tax credits at a higher rate than the current base case policy;
- b. uprating income tax and National Insurance contribution (NIC) thresholds at a rate higher than the current base case policy;¹
- c. cutting income tax (and NIC) rates by some common proportion.

Spending the total revenue gain would, by definition, be sufficient to uprate all taxes and benefits by earnings. Targeting the whole amount on one or other of scenarios (a) and (b) would result in uprating some elements by more than the growth of incomes generally. This may be desirable – either from the perspective of making the benefit system more generous or from the perspective of reducing tax burdens – but would amount to a 'structural change' in the terms we are using in this report. So, for illustrative purposes, we 'spend' just part of the total. We consider what would be possible if 43 per cent of the total were to be available after 20 years (£20.2 billion per year). In other words, that which could be achieved if only three-fifths of the base case revenue gain was actually used for other purposes. The proportion is calculated to correspond to the amount of revenue attributable to fiscal drag alone.² We term spending this proportion of total revenue as 'public finance Scenario 2' with the base case providing 'Scenario 1'.

We examine the impact of using the revenue released under Scenario 2 either: (a) to mitigate the consequences of benefit erosion by financing a higher level of uprating than is applied in the base case; or (b) to increase tax thresholds; or (c) to reduce tax rates.

In order to provide some perspective to the size of these effects, we also consider at various points in our discussion some specific policies into which all the revenue could hypothetically be targeted, rather than more generally applied. These are explained in Box 5.

Box 5 Targeted policies

Three specific policies into which all the revenue could hypothetically be targeted are an increase in the level of:

- per child Child Tax Credit payment;
- Child Benefit for each child;
- the Basic State Pension.

The child payment in the Child Tax Credit is now seen as the primary instrument for directly increasing the incomes of low-income families with children and as a major component to the strategy to reduce child poverty (Hirsch, 2006). Here we simply consider targeting all resources on the child rate, while allowing the threshold to be eroded along with the rest of the credit and benefit systems. If we had also allowed the thresholds to maintain their relative value, this would have maintained the number of families who were entitled and would have made the revenue-neutral increase a lot smaller. Thus the particular illustration we have chosen is targeted especially at the lowest-income families and hence is especially effective at reducing child poverty. Other ways of channelling resources through Child Tax Credit to children would have other effects.

Although it is largely outside the scope of this report to compare alternative ways of reducing child poverty (see Bradshaw and Mayhew, 2006; Brewer *et al.*, 2006a; Hirsch, 2006), we do consider one other approach to child support, which is of interest since its effect does not depend on the way other components of the system are uprated. We explore how large an increase in Child Benefit could be achieved with £20.2 billion to spend and what impact that would make on child poverty.

The Basic State Pension is due to start rising with earnings in 2012 (or by the end of the next Parliament at the latest). However, it is of interest to see what would happen both to pensioner poverty and to reliance on means-tested payments by pensioners if this growth were accelerated. In fact the amount by which the Basic State Pension could be uprated after 20 years of fiscal drag under public finance Scenario 2 would almost exactly restore its 1979 value relative to earnings.³

Table 12 shows the increase in these three payments that would be possible on a first-round, revenue-neutral basis. The increases are shown relative to the 2006–07 starting point value, not the eroded value after 20 years.

The effects across the income distribution on poverty, work incentives and receipt of means-tested benefits are given at relevant points in the text.

Table 12 Revenue-neutral policy changes under public finance Scenario 2after 20 years

Percentage increase in						
Child Tax Credit child amount	152					
Child Benefit for each child	323					
Basic State Pension	63.3					
Changes are relative to the 2006–07 starting point amounts. Revenue neutrality is calculated						

Changes are relative to the 2006–07 starting point amounts. Revenue neutrality is calculated before any behavioural responses.

Source: POLIMOD using FRS 2003–04.

The size of the revenue-neutral change is estimated in a first-round sense – without accounting for any behavioural reactions that may take place. In the case of benefit uprating, *all* benefit and tax credit thresholds, disregards and payment amounts are uprated by the same annual proportion over and above the base case, up to the equivalent of annual earnings uprating. The factor that uses up the revenue under Scenario 2 after 20 years is an increase of 1.87 per cent each year.⁴ So, for a component that is price uprated in the base case, the uprating is somewhere between the base case and what would happen under earnings uprating (assumed to be 2 per cent on top of price indexation per year). This is similar to uprating benefits by a weighted average of prices and earnings growth as used for pensions in Hungary, Finland, Switzerland and other countries (see Chapter 3). For elements that are frozen under the base case, the values still fall in relative earnings terms but by less than they would under the base case.⁵

Spending the revenue under Scenario 2 on uprating tax thresholds is equivalent to uprating by earnings and reducing fiscal drag to zero. The additional factor in this case is therefore 2 per cent.

Spending the revenue under Scenario 2 on cuts in tax rates (including rates of employee and self-employed NICs) would allow them to be cut proportionately by 11.9 per cent after 20 years. This corresponds, for example, to a cut in the basic rate of income tax from 20 to 17.6 per cent.

Figure 20 shows the distributional effects of the combination of fiscal drag and benefit erosion, offset by each of the three changes described above after 20 years. The bars show the effect of the base case (as in Figure 18 in Chapter 8) and the lines plot the net income change (compared with the same starting point) under Scenario 2 with the revenue spent on the three options (a), (b) and (c).

Figure 20 Distributional effects of base case uprating and alternative spending under Scenario 2 after 20 years



Decile groups are defined according to the household disposable income of individuals at the starting point, equivalised using the modified OECD scale. Source: POLIMOD using FRS 2003–04.

Tax 'cuts'

As might be expected, both mechanisms based on reducing tax burdens would have their main effect among higher-income households, with the reduction in tax rates having a slightly greater effect in the top decile group to the extent that the group as a whole would actually gain on average. The value of the tax cut would exceed losses under the base case.

Nevertheless, the political advantages in cutting tax rates make this scenario a not implausible outcome, if governments are committed to keeping the overall tax ratio constant. The direct tax system would become less progressive as thresholds fell in relative terms, but tax rates were cut. On the other hand, as shown in Table 13, poverty rates would rise relative to the starting point as a result of benefit erosion, as the two reductions in tax and contributions have rather little effect on poverty rates.

	Starting point		Spending under Scenario 2 on		
		Base case	Benefit uprating	Increasing tax thresholds	Reducing tax rates
BHC median (£)	363.9	333.8	348.5	344.2	342.7
Poverty rates: all	16.6	23.4	17.8	23.9	24.1
Children	17.5	32.8	20.2	33.1	33.3
Pensioners	22.9	24	22.4	26.2	26
Working age	14.4	19.6	15.6	19.8	20.0
Overall median poverty					
gap (%)	18.6	22.1	18.0	22.9	22.9
AHC median (£)	311.1	279.6	295.4	291.2	289.1
Poverty rates: all	20.6	25.4	22.0	26.4	26.4
Children	26.7	39	29.9	39.1	39.1
Pensioners	16.0	14.3	15.6	19.1	18.7
Working age	19.8	23.7	20.9	23.9	24.1
Overall median poverty					
gap (%)	22.9	35.3	24.1	35.2	35.1

Poverty rates are calculated as the number of people living in households with equivalised income below 60 per cent of the within-scenario median. The modified OECD equivalence scale is used with before housing costs (BHC) incomes and the 'companion' scale is used with after housing costs (AHC) incomes (see DWP, 2007a, p. 189).

Source: POLIMOD using FRS 2003-04.

Benefit 'increases'

In contrast, the combination of the base case mitigated by increased benefit uprating after 20 years shows an almost distributionally neutral effect, with an average loss of around 5 per cent of income across the whole distribution. In other words, if the other

pressures on the public finances necessitated this kind of revenue gain, a balanced way of raising the revenue would be to allow the fiscal drag to occur, but to uprate benefit and tax credit rates in real terms by the factor that would leave only slow benefit erosion (as benefits would be increased annually by 1.87 per cent over 20 years).

Nevertheless, poverty overall would be a little higher after 20 years of Scenario 2 with more generous benefit uprating than under the starting point (Table 13). While pensioner poverty would be a little lower after 20 years of this scenario, child poverty would be higher – 30 per cent compared with 27 per cent on an AHC basis and 20 per cent compared with 18 per cent on a BHC basis. Enhancing the annual rate of benefit uprating by 1.87 per cent would prevent dramatic increases in child poverty on the scale implied by base case uprating, but would not be quite sufficient to prevent child poverty from rising at all.

Targeted benefit or tax credit increases

The effect of uprating benefits and credits across the board by an amount between base case uprating and earnings can be contrasted with the effect of spending the same amount on a single benefit. The distributional effects of the three illustrative targeted ways of spending the resources released under Scenario 2 (see Box 5 earlier in this chapter) are shown in Figure 21.

The very large increase in CTC that could be financed by 20 years under Scenario 2 (increasing its value relative to the starting point by about 1.5 times) would have a strongly positive effect in the second and third decile groups, to the extent that the second decile group of households would be net gainers. As explained in Box 5 earlier in this chapter, increasing the child amounts while letting the income thresholds erode would focus the effect of CTC on households in a narrower band of incomes than is currently the case.

The distributional effect of a Child Benefit increase reflects the fact that the increase would be much smaller for those who receive it and that it would be spread over children at all income levels. The distributional effect is directly related to the position of children in the income distribution.


Figure 21 Distributional effects of targeted spending of the revenue from fiscal drag after 20 years

Decile groups are defined according to the household disposable income of individuals at the starting point, equivalised using the modified OECD scale. Source: POLIMOD using FRS 2003–04.

The effects of increasing the Basic State Pension are also less clearly targeted by income than the CTC increase, but nevertheless would have the greatest proportional effect at the bottom of the distribution, roughly halving the loss experienced by the second decile group from 20 years of base case uprating and almost eliminating that of the bottom decile group.

Poverty and targeted increases

Not unexpectedly, given its distributional impact, the Child Tax Credit increase would have a dramatic effect on the child poverty rate, as shown in Table 14. It would fall to nearly half its AHC starting level (14 per cent compared with 27 per cent) and a third of the base case if all the resources from 20 years of Scenario 2 were used to raise the amount per child. On a BHC basis, the reduction would be to 9 per cent from 33 per cent under the base case (and 18 per cent at the starting point). Putting the

same resources into Child Benefit would reduce the child poverty rate by less, to 27 per cent AHC, more or less the same as at the starting point. However, the poverty gap overall would be lower with the increase in Child Benefit than with the Child Tax Credit increase – 22 per cent of the poverty line rather than 30 per cent. This is due to the fact that, while Child Benefit enjoys almost 100 per cent take-up, this is not the case for the tax credits.⁶ The increase in Child Benefit would not be large enough to pull many poor families across the poverty line, but would reach more very poor families than would the Child Tax Credit increase.⁷

			Spendin	g under Scenario	2 on
	Starting	Base	Child Tax	•	Basic State
	point	case	Credit	Child Benefit	Pension
BHC median (£)	363.9	333.8	350.5	352.1	350.9
Poverty rates: all	16.6	23.4	15.5	19.1	21.9
Children	17.5	32.8	9.3	17.7	35.8
Pensioners	22.9	24.0	27.3	27.5	8.0
Working age	14.4	19.6	14.2	17.0	21.1
Overall median poverty					
gap (%)	18.6	22.1	22.9	19.6	24.6
AHC median (£)	311.1	279.6	298.7	299.1	299.8
Poverty rates: all	20.6	25.4	17.7	22.7	25.4
Children	26.7	39.0	13.5	26.5	41.5
Pensioners	16.0	14.3	20.6	20.8	5.9
Working age	19.8	23.7	18.3	21.8	25.5
Overall median poverty					
gap (%)	22.9	35.3	30.0	25.5	38.3

Table 14 Relative poverty in the UK after 20 years of base case uprating and targeted spending

Poverty rates are calculated as the percentage of people living in households with equivalised income below 60 per cent of the within-scenario median. The modified OECD equivalence scale is used with before housing costs (BHC) incomes and the 'companion' scale is used with after housing costs (AHC) incomes (see DWP, 2007a, p. 189).

Source: POLIMOD using FRS 2003-04.

As well as the effect at a point in time, the different scenarios would have distinct effects on income for any particular family as time went by and children grew to adulthood. Box 6 at the end of this chapter illustrates the case of a low-income couple who have two children and shows, using plausible assumptions about fertility, employment and education, how family income would move relative to the poverty line under alternative uprating scenarios. Putting all the resources into raising the Basic State Pension would have a very dramatic effect on the pensioner poverty rate, reducing it to 6 per cent on an AHC basis, compared with 16 per cent under the starting point. This may seem surprising since low-income pensioners receive the Pension Credit, which is in any case uprated by earnings. The explanation lies partly in the fact that the increase in the Basic State Pension – an increase greater than earnings growth – would take some pensioners above Pension Credit guarantee level and partly in the fact that non-take-up of Pension Credit limits its effectiveness in preventing pensioner poverty. Those not taking up Pension Credit but with entitlements to the Basic State Pension would benefit from the increase in full.¹⁰

Reliance on means-tested benefits and work incentives

The extent to which raising the Basic State Pension, as in this scenario, would reduce reliance on Pension Credit is shown in Table 15. The proportion of pensioners in receipt would fall from 32 per cent at the starting point to 10 per cent. Reliance on any means-tested benefit would fall from 45 to 31 per cent. The proportion affected by Housing Benefit or Council Tax benefit would fall by less – from 21 to 19 per cent. Increasing Child Benefit would not have a corresponding effect on the proportion of children in families in receipt of in-work means-tested benefits, simply because Child Tax Credit does not take Child Benefit receipt into account. Incomes in families with children would rise accordingly. However, increasing Child Tax Credit would, of course, increase dependence on means-tested benefits. The proportion of children affected would be 38 per cent compared with 23 per cent in the base case. This is much smaller than at the starting point (49 per cent) because of the incidence of Child Tax Credit becoming more concentrated as the income thresholds erode.

Spending the resources on benefit uprating across the board increases the number of people facing marginal effective tax rates (METRs) in excess of 50 per cent (Table 16). This would rise to 8.9 per cent from 6.5 per cent of the working population. The increase in Child Tax Credit would make a bigger difference. METRs on average rise by 2.8 percentage points compared with the base case, with the percentage of those with earnings on very high rates (in excess of 50 per cent) rising to 12.4 per cent. Targeting means-tested payment increases on a group of children who are increasingly concentrated near the poverty line because of erosion of the tax credit threshold would have very large effects on the child poverty rate relative to the base case without these increases. But this would be at the cost of increasing METRs on earnings.

			Spendin	g under Scenario	2 on	
	Starting point	Base case	Child Tax Credit	Child Benefit	Basic St Pensic	tate on
All on any benefit or credit*	43	30	35	37	30	27
Children on CTC with IS	17	16	16	16	16	16
Children on CTC alone or						
with WTC**	49	23	32	38	23	23
Pensioners on PC***	32	33	31	33	33	10
Pensioners on any benefit	45	45	45	45	45	31
All on HB/CTB	21	21	21	21	21	19

Table 15 Proportion of people in families in receipt of means-tested benefits or credits at the starting point, after 20 years of base case uprating and under alternative policy changes (%)

* Income Support (IS), Housing Benefit (HB), Council Tax Benefit (CTB), Pension Credit (PC), Child Tax Credit (CTC), Working Tax Credit (WTC).

** Not including cases receiving the CTC family element only.

*** Either Guarantee Credit or Savings Credit, or both.

Source: POLIMOD using FRS 2003-04.

Table 16 Distribution of individual marginal effective tax rates (METRs) after 20 years of base case uprating, and under alternative policy changes: percentage of people in work

			Spendi	ng under Scena	ario 2 on
Range of METR %	Starting point	Base case	Benefit uprating	Tax rate cuts	CTC increase
0 to 19.99	16.7	12.7	12.0	18.9	11.6
20 to 29.99	30.7	25.4	24.2	48.7	23.6
30 to 39.99	31.9	31.0	30.1	24.5	28.2
40 to 49.99	11.2	24.4	24.8	2.3	24.2
50 to 69.99	5.5	3.2	4.7	3.2	6.6
70 to 89.99	3.3	2.5	3.2	1.6	4.8
90+	0.7	0.8	1.0	0.8	1.0
All	100.0	100.0	100.0	100.0	100.0
Mean METR (%)	31.8	32.7	34.3	28.8	35.5

Marginal effective tax rates are calculated for each individual in paid work for a one hour per week increase in gross earnings, taking account of all effects on household disposable income.

Source: POLIMOD using FRS 2003-04.

Box 6 Growing up and out of poverty? Model childhoods

How does fiscal drag and benefit erosion affect the income of low-income families with children over the duration of their childhood? Our projections illustrate the effects of base case uprating on incomes in relation to the poverty line as children get older and are joined by siblings. We then compare them with the effects of some of the alternatives discussed in this chapter. The methodology uses hypothetical illustrative cases based on families who have their first child in 2006 and draws a profile of a 'model childhood' over the next 20 years based on the Lifetime Opportunities and Incentives Simulation model (LOIS).⁸

In general, the arrival of children increases consumption costs considerably. But these are not the only 'costs of a child' because children reduce maternal employment through a combination of periods of absence from the labour market, periods of part-time work and the resulting wage penalties. The 'opportunity costs' in combination with the direct costs change as children age – babies and young children *constrain* more and teenagers and older children *cost* more.

The 'model childhood' that we consider as an illustration is a stereotypical twoadult, two-children family, but with a high likelihood of poverty, as the parents work for the minimum wage. The eldest child is born in 2006 and its younger sibling three years later. Our starting assumption at the birth of the first child is that the father works 40 hours a week and the mother does not work at all until the second and youngest child reaches primary school age, when she takes up part-time work of 20 hours a week. When the youngest child reaches secondary school age, her employment increases to 40 hours a week. When the oldest child reaches 18 and leaves secondary school, both children are also assumed to have left the household.

Figure 22 summarises the underlying assumptions of the 'needs' of this model family as implied by the underlying equivalence scales used in measuring poverty after housing costs (the companion modified OECD scale) and the history of parental employment – shown by the number of hours worked and the childhood and employment events that underlie the employment history. It shows the increasing level of needs on the arrival of the second child and the subsequent ageing of children as the OECD equivalence scale assumes they reach the adult needs level at age 14. In 2025, the eldest child reaches 18 and is no longer in the household; needs are lower in the final two years. The solid line shows the assumptions about hours of work (measured against the left-hand axis).



We explore the evolution of family income and the consequential risk of poverty over 20 years of childhood under the base case uprating assumption. We also consider the difference made by two of the alternative regimes under public finance Scenario 2, discussed elsewhere in this report:

- spending on benefit uprating across the board;
- spending on increasing the child amount in the Child Tax Credit.

Figure 23 shows the comparison of income and poverty for this family using AHC income and comparing it with the poverty line, generated by POLIMOD and applied for the particular family composition at each point in time. The figure shows the interaction of the profile of 'events' and step-changes in needs with the evolving income profile. In the base case scenario, incomes during early childhood (the period of single full-time earnings) declines over time. At the point in childhood where the mother returns to part-time work, there is a clear upward step-change in income in relation to the poverty line, but once again, under the base case, income not only is below the poverty line but also declines relative to it as the tax-benefit package falls in value relative to earnings.



In the later period of childhood, when both children are at secondary school and the mother is working full-time, income is maintained steadily at the poverty line under the base case. This is followed by two step-changes in needs as the younger child reaches 14 and thus has needs equivalent to an adult and the subsequent 'ageing out' of the oldest child when they reach 18 and, by assumption, leave home. The first of these step-changes has a larger negative effect than the positive gains from the mother moving to full-time employment. The second step-change results in income rising clearly above the poverty line.

Indexation is clearly crucial to combat relative poverty over childhood. Combining single full-time, minimum wage work and two children results in an income that is below poverty levels. Base case indexation worsens this position and causes early pre-school childhood family income to drift further into poverty. Only when the older child is no longer present does the one-child couple where both work full-time have income clearly above the poverty line.⁹ Indexation by more than under the base case helps to reduce the decline in incomes, but not to halt it, up to the point where the younger child starts secondary school and the mother takes up full-time work. By that time, the family is entitled to no means-tested tax credits and is in fact slightly worse off than under the base case because they pay slightly more in income tax and National Insurance contributions.

Targeting substantial extra spending on Child Tax Credit would overcome most of these problems over childhood and would lead to growing clearance of income above the poverty line until the increased needs of teenage children took effect and then entitlement was reduced by the older child reaching 18.

Summary

- Letting base case uprating continue unchecked for 20 years would have a significant positive effect on the government budget as a proportion of GDP. Spending a proportion of the budgetary gain could mitigate some of the adverse effects of benefit erosion or fiscal drag. As an illustration, we show what would happen if the required revenue gain was less (57 per cent of it) than that generated by base case indexation and if the remainder was spent in various ways. This is equivalent to foregoing the gains from fiscal drag (and enough to keep the tax ratio constant, other things being equal).
- Spending on *uprating tax thresholds* would benefit mainly households with high incomes and would have little effect on poverty rates. Using the resources to *cut tax rates* would benefit households with top incomes to an even greater extent and would also leave the increase in poverty no lower than in the base case. This scenario is equivalent to allowing fiscal drag to occur, but then reducing tax rates to keep the overall tax ratio constant.
- On the other hand, putting the same resources towards *uprating benefits* over and above what happens under the base case would reduce poverty rates to levels between those at the starting point and under the base case. For pensioners, the poverty rate would be returned to the level at the starting point.

- The combination of the base case mitigated by increased benefit uprating shows an almost distributionally neutral effect after 20 years, with an average loss of around 5 per cent of income across the whole distribution. If the other pressures on the public finances necessitated this level of budgetary gain, a balanced way of raising the revenue would be to allow the fiscal drag to occur, to leave tax rates unchanged, but to uprate benefit and tax credit rates in real terms by the factor that left only slow benefit erosion.
- If, on the other hand, more revenue was required to meet other demands, then a
 distributionally neutral solution (in the sense of all income groups facing the same
 proportional income reduction) would require taxes to be raised such that higherincome households contributed their share.
- Compared to general uprating of benefits and credits, targeting all the spending relative to the base case on *increases in the Child Tax Credit child amount* would have its main impact in the lower part of the income distribution and only among households with children. It would reduce the AHC child poverty rate to half the starting level and almost one-quarter of the poverty rate under the base case over 20 years. It would, however, double the number of parents facing very high marginal deduction rates to 12.5 per cent, compared with 6.5 per cent under the base case.
- Spending the same resources on *increasing the Basic State Pension* would reduce the AHC pensioner poverty rate to 6 per cent, compared with 16 per cent at the starting point. While low-income pensioners are entitled to the Pension Credit, which is uprated by earnings, non-take-up of Pension Credit limits its effectiveness in preventing pensioner poverty. Furthermore, the increase in basic pension – an increase greater than earnings growth – takes many pensioners above Pension Credit guarantee level. The proportion in receipt would fall from 32 to 10 per cent.

10 The older population and uprating policy

The analysis in this chapter investigates the effect of alternative uprating policies on the evolution of pensioner incomes. A particular focus on older people is interesting for several reasons. First, older people are affected by uprating for long periods of time. In retirement, the extent to which one's income keeps pace with or lags behind the incomes of the general population is heavily dependent on how state pensions and other pensioner benefits are uprated each year. Second, the uprating of pensioner benefits has been the subject of considerable debate ever since the Basic State Pension ceased to be linked to earnings and has been changed in recent years. The means-tested minimum income available to pensioners has been increased by at least earnings growth since 1999 and there is now a commitment to link the Basic State Pension to earnings from what is expected to be 2012. In addition, the 2007 Pensions Act changes the way certain parameters of the Pension Credit are uprated as a means of bringing about a structural reform to this benefit. These changes provide an interesting case study.

The analysis uses a dynamic microsimulation model, CARESIM. This model starts with a sample of today's older population. Like POLIMOD, it simulates income tax liabilities, means-tested benefit entitlements and then calculates the net incomes of sample members under different uprating policies. CARESIM also 'ages' the sample through time and so can be used to estimate the future incomes of those members of the sample likely to be alive in six and 20 years' time. This 'ageing' involves: predicting whether each sample member is alive in six and 20 years' time based on official projections of life expectancy (Government Actuary's Department, 2005); projecting the primary incomes of each living sample member taking account of the effect of the death of a partner and making assumptions on indexation of private pension income, etc.; and then simulating tax liabilities, entitlement to, and take-up of, means-tested benefits. Further details are given in Appendix 4.

To understand the impact of uprating policies on the evolution of pensioner incomes, we consider first differences across age groups in the incomes of today's pensioners. Table 17 presents median incomes, poverty rates and receipt of means-tested benefits in 2006–07 by five-year age groups, as simulated by CARESIM. From the youngest to the oldest age group, there is a clear fall in income, while poverty rates and receipt of means-tested benefits increase. Seventeen per cent of those aged 65–69 are poor (BHC) compared with 42 per cent of those aged 85 and over. AHC poverty is lower but the trend across age groups is similar, doubling from 8 per cent

amongst 65–69 year olds to 16 per cent of those aged 85 and over.¹ Nineteen per cent of the youngest age group receive Pension Credit compared with 48 per cent of the oldest. These differences could be due to individual pensioners experiencing reductions in their incomes, or to later generations of pensioners retiring on (and maintaining) higher incomes than earlier ones, or to a combination of the two.²

Age in 2006–07						
	65–69	70–74	75–79	80–84	85+	65+
Median income (£ per week)						
BHC	295	272	258	244	240	266
AHC	289	259	242	230	223	250
Poverty rate (%)						
BHC	17	23	30	38	42	28
AHC	8	11	14	16	16	12
Receiving PC (%)	19	23	31	37	48	29
Receiving CTB (%)	29	34	38	43	47	36
Receiving HB (%)	16	17	20	24	27	20
Receiving any means-tested						
benefits (%)	32	38	43	50	57	42

 Table 17 Incomes, poverty rates and receipt of means-tested benefits among today's pensioners by age group, relative earnings terms

Source: CARESIM using FRS 2003-04.

Box 7 Assumed growth in non-state income and other relevant monetary amounts used in the analysis of pensioner incomes

- Income from non-state pensions increases in line with the Retail Prices Index. If anything, this is likely to overestimate the increase in these sources of income that pensioners receive in practice. Defined benefit private pensions tend to increase by at most price inflation and many pensioners drawing annuity-based pensions opt for annuities that remain at the same nominal level throughout retirement.
- Capital holdings are assumed to remain constant in nominal terms and thus to fall in relative earnings terms. This is equivalent to assuming that pensioners consume all the income from their capital but do not deplete the original capital sum. No allowance is made for windfall increases in capital – for example, from inheritances.
- Since capital is assumed to remain constant in nominal terms, so too is income from it. By implication, interest rates are assumed to remain constant in nominal terms.
- Rent and Council Tax are assumed to increase in line with earnings.

The results of the analysis in this chapter are therefore presented in two ways. First, we consider the impact of policies on the incomes of pensioners who are at least 65 years old in 2006–07 as they age over the next 20 years. This helps us to gauge the extent to which individual pensioners experience changes in their income over time and the role of uprating policies in influencing the path of their incomes. Second, we compare the incomes of today's pensioners with projections of the incomes of people of the same age in six and 20 years' time. For example, the incomes of people currently aged 85+ are compared with projections of the incomes of those aged 85+ in 20 years' time where the latter group are the survivors of those currently aged 65+.

Uprating assumptions for taxes and state benefits used in the base case are those set out in Chapter 5, including the effect of the structural reforms announced in the 2007 Budget. Assumptions on earnings and RPI increases are also the same as in previous chapters. However, there are differences in how primary income from other sources is assumed to change. In earlier chapters, primary incomes are assumed to all grow together in line with average earnings (or prices). Here, we make particular assumptions about each source in order to model how the incomes of pensioners are likely to change in practice. These assumptions are listed in Box 7.

As before, the base case assumptions for uprating policies amount to a continuation of current policy, including the changes contained in the 2007 Pensions Act and the 2007 Budget. The alternative policies examined are as follows.

- 1 The uprating policy for the Basic State Pension (BSP) and Pension Credit (and linked parameters of Housing Benefit and Council Tax Benefit) is that which predates the Pensions Act. The BSP and the Savings Credit threshold are linked to the Retail Prices Index but, as in the base case, the (ordinary) Guarantee Credit level is linked to earnings.
- 2 Price linking (RPI or ROSSI as appropriate) of all components of state pensions, Pension Credit and HB and CTB for pensioners, except the historically frozen elements such as the capital thresholds.
- 3 As (1) above, but with revenue under Scenario 2 used to increase the Basic State Pension (see Box 5 in Chapter 9).
- 4 As (1) above, but income from the State Second Pension (S2P) and its predecessor, the State Earnings Related Pension (SERPS), is indexed to earnings rather than prices from 2012.

CARESIM ages an initial sample of people aged 65 and over, so these people (if they are still alive) are aged 71 and over after six years, and 85 and over after 20 years. Therefore, when we compare the incomes of today's older people with CARESIM's projections of the incomes of people of the same age in six and 20 years' time, the age groups we can analyse are limited. We therefore focus on those aged 75+ and 85+ for the comparisons between now and six years' time, and on people aged 85+ for comparisons with 20 years' time.

The analysis is also restricted to people who were single and over state pension age in the starting year, or part of a couple where both partners were over state pension age. Income is measured a little differently from Chapters 6 to 9. It is the total income of the benefit unit rather than the household (ascribed to each person in the household). Also disability/care-related social security benefits - Attendance Allowance, Disability Living Allowance and Carer's Allowance and the associated premiums in Pension Credit - are excluded from income. These benefits are intended to provide help with the extra costs associated with disability. If they are included in income, the incomes of disabled people will appear higher than similar people without disabilities when their standard of living is not actually any higher. As receipt of these benefits is much higher among the older population than in the general population, we exclude them from income in the analysis in this chapter. In calculating poverty rates, poverty thresholds are based on the median income levels, in relative earnings terms, given in Table 4 in Chapter 6. Because of differences in the definition of income and of pensioners, pensioner poverty rates and other results are not directly comparable with those in earlier chapters or with official Households Below Average Income (HBAI) statistics.³ However, it is the trends over time and differences across uprating policies that are the main point of interest.

The impact of uprating policies on pensioners as they age

Table 18 and Figures 24, 25 and 26 show the effect of different uprating policies on pensioners as they age. For simplicity, the figure shows only BHC poverty rates but AHC rates are included in the table. Monetary amounts are expressed in relative earnings terms. Under the base case, the poverty rate (BHC) among those who survive the next 20 years rises from just over a fifth (22 per cent) to getting on for a third (30 per cent) after 20 years.

ve afte	
and ali	
6-07	
in 20(
ed 65+	
ole age	
e: peol	
ley ag	
's as th	
sioner	
n pen	
licies (
ing po	
uprati	
rnative	terms
of altei	rnings
npact (ive ea
18 Ir	, relat
Table	years

	2006–07 starting		A	fter six yea	ſS			4	tter 20 yeaı	Š	
	point	Base case	Pre-2007 Pensions Act	Prices uprating	Spending under Scenario 2 on BSP	SERPS/ S2P linked to earnings from 2012	Base case	Pre-2007 Pensions Act	Prices uprating	Spending under Scenario 2 on BSP	SERPS/ S2P linked to earnings from 2012
Median income (£ per	veek)				000		ľ.	L		L	
BHC	282	203	282	292	282	263	231	C52	193	G67	244
AHC	263	251	265	239	272	250	227	232	179	295	232
Poverty rate (%)											
BHC	22	25	23	30	16	24	30	29	54	5	26
AHC	10	1	10	21	7	10	10	12	45	0	8
Receiving PC (%)	23	30	33	20	21	30	35	50	19	7	33
Receiving CTB (%)	32	36	37	31	30	36	41	47	33	24	40
Receiving HB (%)	16	16	17	16	16	16	17	18	16	13	17
Receiving any MTB (%) 35	41	43	34	35	41	46	57	37	27	45



Figure 24 Effect of alternative uprating policies on people aged 65+ in 2006–07 and alive after 20 years: poverty rates BHC

The percentage receiving Pension Credit increases from 23 to 35 per cent, while receipt of any means-tested benefit rises from 35 to 46 per cent. Poverty rates rise a little more after the Pension Act, but rates of receipt of means-tested benefits rise much less than if previous uprating policy had continued. Under pre-Pension Act policies, half of this generation of pensioners would be receiving Pension Credit after 20 years, with the proportion on any means-tested benefit reaching 57 per cent.

The 'base case' and pre-Pensions Act results already incorporate the current Government's commitment to earnings uprating of the Guarantee Credit. In the context of the discussion elsewhere in this report of the effect of the lack of such a commitment for other age groups, it is instructive to look at what would have been implied if the means-tested minimum for pensioners had instead remained price linked. The results show that a return to price indexation of all but the frozen elements of pensioner benefits would contain the growth in Pension Credit receipt. After 20 years, the proportion receiving Pension Credit would be slightly lower than at the start of the period (19 compared with 23 per cent) and the proportion receiving any means-tested benefit would be only a little higher (37 compared with 35 per cent). But this would be at the cost of a poverty rate in 20 years' time of close to onehalf (54 per cent BHC; 45 per cent AHC).





Going in the other direction, in contrast, spending the revenue under Scenario 2 on increasing the Basic State Pension would reduce poverty to very low levels and substantially diminish dependence on means-tested benefit. Poverty among this generation of pensioners would fall to 5 per cent (BHC), receipt of Pension Credit to 7 per cent and receipt of any means-tested benefit to 27 per cent.

Linking the S2P/SERPS to earnings from 2012 would reduce future poverty rates and dependence on means-tested benefits by slightly more than under the base case.

Figure 27 provides a gender breakdown of the effect of alternative uprating policies on poverty (BHC – income is still measured as the income of the benefit unit). Women in this generation are more likely to be in poverty than men at the start of the period, and also more likely to be receiving means-tested benefits. This remains true for future years under each of the alternative uprating policies. None affects one sex more than the other.



Figure 26 Effect of alternative uprating policies on people aged 65+ in 2006-07 and alive after 20 years: receipt of any means-tested benefit

The impact of uprating policies on differences in the incomes of current and future pensioners

In six years' time, poverty rates and reliance on means-tested benefits among pensioners aged 75+ and 85+ would be similar to today under the base uprating policy and also under pre-Pension Act policies or if S2P/SERPS was linked to earnings from 2012 (Table 19). In 20 years' time, poverty rates would be lower among those aged 85+ than today under each of these three uprating policies (Figures 28, 29 and 30). Just over two-fifths (42 per cent) of people currently aged 85+ are poor (BHC). In 20 years' time, the equivalent proportion would be 30 per cent under base case policies and a little lower under pre-Pension Act policies or linking S2P/SERPS to earnings. Receipt of Pension Credit would have fallen from just under a half (48 per cent) to 35 per cent under the base case but before the Pension Act reforms would have risen slightly.





Source: CARESIM.

Table 19 Impact of a 20 years' time, relative	Ilternative ve earning	e upratir gs term	ng policies s	:: compar	ison of to	oday's pens	sioners v	vith those	of the sa	ime age ir	six and
	2006–07		A	tter six year	S			4	After 20 year	ý	
	point	Base case	Pre-2007 Pensions Act	Prices uprating	Spending under Scenario 2 on BSP	SERPS/ S2P linked to earnings from 2012	Base case	Pre-2007 Pensions Act	Prices uprating	Spending under Scenario 2 on BSP	SERPS/ S2P linked to earnings from 2012
Aged 65+ median income (£ per w∈ BHC AHC	3ek) 266 250										
Poverty rate (%) BHC AHC	28 12										
Receiving PC (%) Receiving CTB (%) Receiving HB (%) Receiving any MTB (%)	20 20 20 20										
Aged 75+ median income (£ per w∈ BHC AHC	зек) 250 232	248 234	250 238	235 214	265 251	248 234					
Poverty rate (%) BHC AHC	36 15	32 32 14	31 31 13	38 27	8 23 23	32 32 14					
Receiving PC (%) Receiving CTB (%) Receiving any HB (%) Receiving any MTB (%)	37 23 49	37 21 48	40 21 43	28 37 20	28 36 42	37 21 48					
Aged 85+ median income (£ per we BHC AHC	эек) 240 223	233 220	236 225	218 197	250 233	234 220	237 227	234 232	193 179	295 296	244 232
										Continu	ied overleaf

After six years			٩	fter 20 year	Ś	
Spendinç under	g SERPS/ S2P linked		Pre-2007		Spending under	SERPS/ S2P linked
Prices Scenario uprating 2 on BSF	to earnings from 2012	Base case	Pensions Act	Prices uprating	Scenario 2 on BSP	to earnings from 2012
47 32	41	30	29	54	ъ	27
36 10	16	10	12	45	N	8
38 38	48	35	50	19	7	33
44 43	49	41	47	33	24	40
25 25	26	17	18	16	13	17
51 52	56	46	56	37	28	45
(D)	Brices Scenario uprating 2 on BSF 47 32 36 10 38 38 38 38 44 43 25 25 51 52	anices Scenario to earnings uprating 2 on BSP from 2012 47 32 41 36 10 16 38 38 48 44 43 49 25 25 26 51 52 56	Section Detrined between Detrined between Detrined between Detrined between Descention uprating 2 on BSP from 2012 case case 47 32 41 30 33 36 10 16 10 35 38 38 48 35 41 25 25 26 17 56 51 52 56 46 46	Section Section <t< td=""><td>Section Scenario to earnings Base Pensions Prices Prices uprating 2 on BSP from 2012 Base Pensions Prices 47 32 41 30 29 54 36 10 16 10 12 45 38 38 48 35 50 19 44 43 49 41 47 33 25 25 26 17 18 16 51 52 56 46 56 37</td><td>sPricesScenarioLite-LouLite-Louuprating2 on BSPfrom 2012BasePensionsPricesScenariouprating2 on BSPfrom 2012caseActuprating2 on BSP4732413029$54$$5$3610161012$45$$2$38384835$50$19$7$44434941$47$$33$$24$252526171816135152564656$37$$28$</td></t<>	Section Scenario to earnings Base Pensions Prices Prices uprating 2 on BSP from 2012 Base Pensions Prices 47 32 41 30 29 54 36 10 16 10 12 45 38 38 48 35 50 19 44 43 49 41 47 33 25 25 26 17 18 16 51 52 56 46 56 37	sPricesScenarioLite-LouLite-Louuprating2 on BSPfrom 2012BasePensionsPricesScenariouprating2 on BSPfrom 2012caseActuprating2 on BSP4732413029 54 5 3610161012 45 2 38384835 50 19 7 44434941 47 33 24 252526171816135152564656 37 28

Table 19 Impact of alternative uprating policies: comparison of todav's pensioners with those of the same age in six and



Figure 28 Effect of alternative uprating policies on people aged 85+ in 2006–07, 2012–13 and 2026–27: poverty rates (BHC)

If we were to revert to general price uprating, the model shows poverty rates rising even after six years. The increase would be modest for those aged 75+ (from 36 to 38 per cent, BHC) but by more for people aged 85+ (from 42 to 47 per cent). In 20 years' time more than half (54 per cent) of those aged 85+ would be poor under prices upratings. But, as before, prices uprating would reduce the proportions receiving Pension Credit and means-tested benefits in general. In contrast, spending the revenue from fiscal drag on increasing the Basic State Pension would reduce poverty rates – very substantially after 20 years – and reliance on means-tested benefits.

This chapter has shown that linking the Guarantee Credit and Basic State Pension to earnings will substantially reduce pensioner poverty compared to general price uprating. However, there remain parts of the pension system that are not linked to earnings, or where practice has been for them to remain frozen in nominal terms for long periods. Box 8 at the end of this chapter provides illustrations for some particular cases.



Figure 29 Effect of alternative uprating policies on people aged 85+ in 2006–07, 2012–13 and 2026–27: receipt of Pension Credit

Figure 30 Effect of alternative uprating policies on people aged 85+ in 2006–07, 2012–13 and 2026–27: receipt of any means-tested benefit



Source: CARESIM.

Box 8 Living through retirement and the effects of uprating

How poorer retirees will be protected against poverty through their retirement is illustrated by a pair of newly retired pensioner couples.⁴ They both have private pension incomes, additional savings and own their homes outright. To make commentary and discussion easier we give names to the two couples. The Chaudhuris, both aged 65, have full state pensions and both have also paid into their employers' defined benefit occupational pensions. Between them they have £81 a week income from these pensions and these are assumed to keep pace with inflation. Their neighbours, the Davies, have exactly the same circumstances, but their private pensions are from personal pension schemes that give them £81 per week in an annuity that is not price protected. Both couples have £18,250 in savings and pay £17.90 per week Council Tax.

Figure 31 shows how income evolves in relation to the BHC poverty line for these couples over the next 20 years of retirement – until they all reach the age of 85. The poverty line is taken from earlier analysis in Chapter 6 and reflects the combined impact of benefit erosion and fiscal drag on median income. The two couples begin their retirement with the same level of income. This is less than 20 per cent higher than the BHC poverty line in spite of having substantial private pensions. Income declines for both couples, but the Davies' falls faster because their private annuity pensions are not protected against inflation, to the extent that they slide into poverty in 2015 (on our assumption of no other changes – for example, to their health).

This happens because of a combination of fiscal drag on Council Tax and the effect of their savings of £18,250 on Council Tax benefit entitlement. Council Tax rising with earnings takes a higher proportion of their incomes over time. However, they can never claim Council Tax benefit to assist them, as their savings are over the £16,000 capital limit, which is never uprated. They are not protected by the Guarantee Credit, which provides a safety net income at or around poverty income levels. If they were, then their capital would no longer exclude them from entitlement to help with Council Tax. However, income would not fall to Guarantee Credit level within the 20 years we consider.

The prospects for both couples would improve when their private pensions fell further and their retirement income fell to Pension Credit levels. At this point, a further six more years on (in 2032 when they are aged 91), their incomes would rise significantly as they would also have their Council Tax paid in full.



This illustrates how the combination of capital rules and uprating policy can result in strong disincentives to maintain even the nominal value of savings in retirement for those with moderate private incomes. Both couples would be well advised to draw from their capital and, say, take the 'cruise of a lifetime', which costed \pounds 6,000 in total, on their 66th birthdays. If this amount were taken from their savings, this would result in neither couple ever sliding across the poverty line.

Summary

Under current policies, pensioners face an increasing risk of poverty and of reliance on means-tested benefits as they age. However, the 2007 Pension Act reduced substantially the growth in the likelihood of receiving Pension Credit. Both the Pensions Act and the uprating practice that immediately predated it substantially reduced the rise with age in the risk of poverty compared with general price uprating. Price uprating would have resulted in a reduced risk of being means-tested as pensioners aged. Diverting revenue from fiscal drag into the Basic State Pension would mean pensioners facing reduced risk of both poverty and reliance on meanstested benefits.

In contrast to the picture for individual pensioners, under base case uprating, we can expect poverty among the oldest pensioners (85+) to be lower in 20 years' time than it is for pensioners of the same age today. As a direct result of the Pension Act reforms to uprating policies, fewer of them will be receiving Pension Credit and means-tested benefits more generally. This could have been achieved by reverting to general price linking but at the cost of rates of poverty exceeding 50 per cent. Spending the revenue from fiscal drag on increasing the Basic State Pension would reduce both poverty and receipt of Pension Credit among those aged 85+ to very low levels in 20 years' time.

Caution needs to be exercised in extrapolating from the projections for the age groups considered here to pensioners is general. This is because, as discussed at the beginning of this chapter, differences in the incomes of different generations of pensioners are at least partly due to differences in the levels and sources of income on retirement. These are harder to predict than the effects of uprating on the incomes of today's pensioners.

11 Conclusions: choices and trade-offs

Much of the debate over the tax and benefits system surrounds structural changes to the system – changes to the tax system introduced in each year's Budget, or reforms to benefits and tax credits preceded by government policy documents, and sometimes requiring legislation. But, as this report has illustrated, the quieter decisions taken each year on how benefit levels and tax thresholds are adjusted for inflation and income growth can, over time, have much larger effects.

In this report, we show the results of modelling what would happen (other things being equal) if we continued uprating benefits, tax credits and direct tax thresholds each year in line with current policies for the next 20 years. Among other effects, one would expect a very considerable, if silent, improvement in the public finances, measured in relation to national income, as a result of what is known as 'fiscal drag' and what we have called 'benefit erosion'. Overall, the budgetary effect could be a benefit to the public finances equivalent to up to 3.6 per cent of GDP after 20 years.

But, at the same time, the incomes of a considerable part of the poorer nonpensioner population would fall behind those of the population as a whole and of those on higher incomes, and relative poverty would rise. For instance, in this 'base case', child poverty as conventionally measured (before housing costs) would rise from 18 per cent at the starting point to 33 per cent after 20 years. Instead of eradicating child poverty, the effect of continuing current uprating policies would be almost to double it. Looking at different family types, Figures 6, 8, 9 and 10 in Chapter 6 illustrate the large deterioration in the ability of the tax and benefit system to protect them from relative poverty.

Even looking only in real terms and at poverty rates against an absolute threshold, the freezing of some elements of the system in nominal terms leads to some groups falling behind, including some groups with low earnings affected by the way in which thresholds for tax credits are calculated (Box 4 in Chapter 6). Nor would such changes necessarily improve work incentives (Chapter 7).

Ad hoc and structural reforms – enabled in part by the creeping revenue gain – can, of course, offset such gradual effects. But the analysis here suggests that such changes would have to be very frequent if they were to do so. For instance, the reforms announced in the 2007 Budget can be expected to have a progressive effect, raising incomes of those in the lower-income groups overall and reducing relative poverty. However, a reform of that scale would be required every two to three years to offset the rise in relative child poverty that benefit erosion implies under current

policy, and indeed every year to offset the fall in relative income for lower-income groups as a whole.

Policy-makers may see fiscal drag and benefit erosion as politically painless or 'stealthy' ways of improving the public finances without taxing incomes at a higher rate. This may, for instance, be seen as necessary to cope with the demands of an ageing society. Figure 18 (Chapter 8) shows how the default evolution of the benefit and tax system would cause people overall to retain less relative to an average earned income, the inference being that Government could take more to spend either on things other than income transfers or on the extra cost of income transfers caused by demographic and other changes.

In general terms, this might be acceptable. As people live longer they may be prepared for their disposable income to form a lower proportion of their average annual (lifetime) gross earnings than it did for their parents, because the difference would be paying for a longer non-working life and/or for more, or higher-cost, state-provided health care. Two particular factors might potentially help to make such an outcome more acceptable. One would be that it would come about gradually, during a time when incomes were growing, and so would not involve actual cuts in living standards. The other might be if this were happening to everyone. If so, for instance, median net incomes – and hence a relative poverty line – would grow more slowly than gross earnings, but relative poverty itself would not necessarily increase.

But the problem is, as demonstrated by Figure 18 in Chapter 8, that achieving a gain to the public finances like this would *not* affect everyone in the same way. The distributional impact would be very unequal. Put simply, 'benefit erosion' has greater impact on people in lower-income groups than fiscal drag, because transfers to and from Government comprise a greater part of income than they do for people in higher-income groups. And this has huge implications for relative poverty, even after taking account of the way median incomes would grow more slowly than gross earnings.

So, while it is possible to argue that fiscal drag and benefit erosion might be necessary, at least to some degree, to finance other things, our analysis highlights the inequitable way in which the burden of doing so would be distributed, and the extent to which we could be on a structural route to growing inequality.

In the event, of course, the tax and benefit system will not be left on autopilot for the next six years, let alone for the next 20. However, structural reforms would not *necessarily* improve the distributional position. For instance, we show in Chapter 9 what would happen if, instead of Government accepting the whole revenue gain implied by fiscal drag and benefit erosion over 20 years, it decided to cut tax rates to offset the rise in the tax ratio otherwise implied by fiscal drag. This is not such an unlikely scenario. Governments might well want to avoid a rise in the tax ratio – this is, after all, assumed in the Treasury's own long-term fiscal projections. Lower tax rates could be seen as the most politically visible tool. But the effects would be regressive. Using results from Chapter 8, Figure 32 contrasts the distributional impact of this route with an alternative scenario – with the same overall net revenue gain to Government, but where benefits and tax credits are uprated annually by an amount closer to earnings growth than prices growth. Under the first scenario, net incomes would fall considerably behind average gross earnings for those in lower-income groups, and relative poverty rises, but those with the highest incomes would actually see their net incomes rise *faster* than earnings as they gained from cuts in tax rates. Under the alternative shown, all income groups would fall behind earnings growth, but would do so in proportion, and relative poverty would not rise nearly so dramatically.¹





Source: POLIMOD using FRS 2003-04.

While we have argued that, in the main, the choices made about year-to-year uprating and indexation are invisible and their effects little noticed, this has been far from the case for pensions policy. Chapter 10 – looking at the prospects for cohorts of current pensioners as they age – illustrates the effects of the series of the controversies and policy decisions stretching over more than 20 years that culminated in the 2007 Pensions Act. While the poverty rate (before housing costs) for today's pensioners aged 85 and older is modelled at 42 per cent, this could have risen to over 50 per cent for the equivalent group in 20 years' time if all pensioner benefits were to be price-linked. By contrast, with the reforms now in place, with the Guarantee Credit and (eventually) the Basic State Pension linked to earnings, the poverty rate for this group would fall to 30 per cent.

But the outcome of those reforms is a sharp distinction in practice between the pensioner and non-pensioner populations. If one returns to the principles outlined in Chapter 2 (and summarised in Table 1), pensioners now have a regime that is more in line with giving a return on contributions, and a little more generous than would be necessary to be consistent with unchanging inequality or relative poverty. By contrast, the *default* treatment of other non-working households is aimed only at preventing falls in their real standard of living, and that of families with low earnings and children, only a little more generous. As we have stressed, ad hoc policy reforms can protect some groups, and have periodically done so, as illustrated in the case of the 2007 Budget reform announcements. But this serves to illustrate that perhaps the main principle being applied is to maintain government flexibility.

As our survey of international practice in Chapter 3 shows, the UK is not alone in the debate being more lively for pensions than other aspects of the system, or in applying a variety of approaches to different groups. There is no 'international best practice' to pull off the peg – for the very good reason that the starting points of countries' systems vary, as do the principles that drive their political priorities.

However, international comparisons of uprating practice highlight the importance of transparency about the nature of default and its implications. The use of 'hybrid' indices in some countries combines the effects of changing prices and incomes in one compromise index. The precise formula must be defined and nature of the compromise discussed, helping to make the political choices explicit. In practice, such a hybrid index could correspond to something like the 'distributionally neutral' regime illustrated by the nearly horizontal line in Figure 32. Or, if the case could be made, it could involve greater reductions to the relative value of benefits and tax credits combined with tax increases, which would maintain distributional neutrality while releasing resources for other priorities. It could also be constructed to favour either rich or poor. The point is that the intention and the effect would be explicit and open for debate.

As things stand in the UK, with the combination of high poverty rates in international terms, particularly for children, and a system largely using price linking as a default, the consequences of leaving decisions about uprating on autopilot are very large and deserve much more open discussion than has been the case.

Notes

Chapter 2

- This is distinct from the problems of actually determining the appropriate flow of resources to tax or take into account in means testing. At times of inflation, real interest receipts or capital gains are much lower than nominal receipts or gains, but it is arguably the former that should be the basis of income assessments. Such 'capital-income adjustment' issues are beyond the scope of this report (Meade Committee, 1978, Chapter 6).
- 2. See also Johnson (1996) who discusses similar issues in relation to pensions only.
- 3. The CPI is an internationally comparable price index used in particular to assess and compare inflation rates across the European Union. It covers somewhat different items from the RPI – for instance, including air fares, university accommodation fees, foreign students' university tuition fees, and unit trust and stockbrokers' charges, while excluding mortgage interest payments, Council Tax, house depreciation, road tax, buildings insurance and various health expenditures.
- 4. At the time of writing in 2007, the 'long term' applies after 2011–12, but, of course, this horizon shifts forward as time passes.
- 5. This was also expected to be the case in the 2008 uprating, as the September 2007 RPI increase (3.9 per cent) was higher than the July 2007 average earnings increase (3.6 per cent).

Chapter 3

 Our sources include OECD (2004, 2005), the Mutual Information System on Social Protection in the EU and European Economic Area (European Commission, 2005), national legislation and budget documents when available in English, and consultation with national experts. We are grateful to Sara Baetens, Didier Coeurnelle, Judi Egerton, Ann Harding, Christopher Heady, Carol Krahe, Natasa Kump, Hanna Nicholls, Satish Parmar, Alari Paulus, Alicia Payne, Daniel Van der Sypt, Peter Whiteford, Eszter Zolyomi and Anthony Zuza.

- 2. According to the formula, if nominal wage growth, measured in the two previous years, exceeds 2 per cent a year, a maximum of 0.3 percentage points of the excess over 2 per cent gets allocated to a social spending reserve.
- 3. For example, in 2007, the rates of Child Care benefit were raised 10 per cent more than price indexation.
- 4. Previously, in the early 1980s, the amount was uprated as a gross (and then a net) proportion of average earnings until, in 1989, CPI became the basis for indexation, provided that the amount stayed between the 65 and 72.5 per cent of net earnings. Amounts were subsequently frozen between 1991 and 1993, when the floor–ceiling band was restored.
- 5. A reform of social assistance benefits is currently under discussion in Slovenia with one of the aims being the unification of different uprating regimes for social benefits (except pensions).
- 6. Also, the uprating index in the overall calendar year cannot exceed the annual growth index of average national salaries and cannot be inferior to CPI unless the annual growth index of average salaries is lower than CPI.
- 7. Indexation is not an issue for the tax schedule because a flat income tax has been in place since 1994. The personal allowance has been increased several times and has more than doubled overall in real terms between 2000 and 2007.

- Since 1977, when the Rooker-Wise amendment to the Finance Act made increasing personal allowances in line with inflation a statutory commitment, indexation with prices (at least) has occurred in most years. This did not happen in 1981–82, 1993–94, 1994–95 or 2003–04 when the main single allowance was frozen.
- 2. HMRC Table 2.7, www.hmrc.gov.uk/stats/t_expenditures/menu.htm.

- 1. In fact, as announced in the 2008 Budget, but not taken into account, CTC child amounts will increase by more than earnings growth in 2008, 2009 and 2010.
- 2. As announced in the 2007 Budget, a tax rate of 10p will continue to apply to people with investment income who would previously have been taxed at that rate (i.e. to the extent that other income does not exceed the equivalent of the former 10p band). Those eligible will have to apply to have their tax deducted in this way and it is not obvious that take-up of this concession will be 100 per cent. For simplicity of modelling, we have not incorporated this aspect of the reform.
- 3. In our results showing the effects of the reform as if it had been implemented all at once in 2006–07, we take future cash values as announced in the 2007 Budget documents and adjust them back to 2006–07 values using the normal indexation factors between the relevant years. Appendix 3 provides details. Thus, for example, Child Benefit for the first child was to reach £20 in 2010. It is assumed that this is in nominal terms and this amount is just 22 pence more than it would be under price indexation. The real value of the increase reduces to 20 pence in 2006–07 terms. As announced in the 2008 Budget, the £20 level for the first child will be reached one year earlier in 2009 and will therefore be worth a few pence more than taken into account in the modelling.
- The 'base case' does not include the additional £25 in maximum CTC announced in the 2007 Pre-Budget Report, nor the further increases announced in the 2008 Budget.
- 5. See note 3 above.

- 1. Note that these estimates, as well as all other POLIMOD results, take account of non-take-up of the means-tested elements of benefits, tax credits and pension incomes. See Appendix 4 for more information.
- 2. HM Treasury (2007a, Table A1).
- 3. Assuming that rents, Council Tax and the minimum wage all grow in line with average earnings, moving as described in Chapter 5. The poverty line is that derived from POLIMOD and given in Table 4 in this chapter.

- 4. The poverty line is derived from the analysis of data on a sample of the whole population, as explained later in this chapter.
- 5. The median poverty gap is the proportion by which the household income of the typical person in poverty falls short of the poverty line.
- On a BHC basis, after six years, the median poverty gap actually falls a little – from 19 to 18 per cent.
- 7. Poverty measured relative to a fixed line that is adjusted only for inflation.
- 8. As explained in Chapter 5, we model this by taking the projected parameters of the tax and benefits systems after six and 20 years, adjusted back to the base year by expected earnings growth, and then applied to the population's distribution of other gross incomes as they were in 2006–07. Note that these estimates do not include the effect of continuing to index the thresholds for *employer* NICs with prices. It is likely that in practice these would move consistently with *employee* NIC thresholds.
- 9. Within these hours limits, this, together with higher income tax and NICs, has a greater effect than the indexation by earnings in the early period of the child amounts in Child Tax Credit.

- 1. Income Support (IS), Housing Benefit (HB), Council Tax Benefit (CTB), Pension Credit (PC), Child Tax Credit (CTC), Working Tax Credit (WTC).
- 2. The proportions are not exactly the same at the starting point and after six and 20 years because of the way the Savings Credit is uprated see Appendix 1. Note that, as explained in Chapter 5, this is not a forecast of how many pensioners would be receiving Pension Credit in 2026–27, as it takes no account of factors such as the changing composition of pensioner income or the ageing of the population. As before, the comparisons isolate the impact of indexation processes by themselves.

- 1. Elements currently uprating using the Rossi index remain treated in this way.
- 2. It is worth noting that also amending the base case assumptions to include earnings uprating of the child CTC amount by earnings throughout the period reduces the child poverty rate on a BHC basis by a further 5 points (to 26 per cent). This is a significant reduction, but does not return the child poverty rate to that at the starting point because the threshold as well as other benefit incomes remain indexed to prices and because the CTC child amount remains just one component of the support received by households with children.

- 1. Not including employer NIC thresholds.
- 2. This should not be taken to suggest that the revenue from fiscal drag should (or indeed could) be ring-fenced in this way. We simply take this proportion of the total revenue in order to provide some sense of scale to an otherwise arbitrary choice.
- 3. Under the assumption of no changes in pensioner characteristics or numbers.
- 4. Elements that are frozen under existing policy (like capital thresholds for meanstested benefits) are uprated each year by this factor alone; elements that are currently uprated by the RPI are uprated by 1.87 per cent more than this annually, meaning that their value falls to 98 per cent of the starting value instead of 67 per cent as under the base case.
- 5. They fall to 56 per cent of their present value after 20 years instead of 39 per cent under the base case.
- 6. See Appendix 4 for a description of how non-take-up is captured in the modelling.
- 7. This depends on the assumption that increases in entitlement do not encourage a higher rate of take-up.
- 8. For more details, see Appendix 4; and Evans and Eyre (2004) and Evans and Scarborough (2006).

- 9. Under all three regimes, after 20 years and when both parents work full-time and there is only one dependent child, incomes are above the poverty line. The distance above varies partly because of differences in taxes and benefits and partly because the poverty line varies across regimes.
- 10. Non-take-up of Pension Credit and other means-tested elements of the system is accounted for in the modelling, as explained in Appendix 4.

- 1. As explained later, these figures are not directly comparable with the poverty rates in the previous chapters or indeed with the official Households Below Average Income statistics. It is the comparison across age groups that is important to note at this point.
- 2. A decline in income across the age groups would also be observed if those on higher income died younger than those on lower incomes. Since all the evidence suggests the opposite richer people live longer than poorer people this is unlikely to be an explanation. Rather, the fall in income across the ages would be more marked if there was less difference in life expectancy between rich and poor.
- 3. The exclusion of disability benefits not only affects income before meanstested benefits but also reduces entitlements to means-tested benefits through the severe disability premium. This will affect BHC but not AHC incomes. In CARESIM, all owner-occupiers are assumed to have paid off their mortgages, which results in smaller differences between BHC and AHC income than in the POLIMOD results. Combined with the use of benefit unit rather than household income, and the exclusion of benefit units in which one partner is under state pension age, these differences explain why, for the 65+ group as a whole, BHC poverty rates are higher and AHC poverty rates are lower than the pensioner poverty rates estimated by POLIMOD.
- 4. These hypothetical examples are informed by the 2005–06 Pensioners' Income series, www.dwp.gov.uk/asd/pensioners_income.asp.
Chapter 11

 The reason it rises at all is that the effect on incomes of the scenario in which benefits are uprated by more than prices is not exactly the same across the income distribution. At the median (4.4 per cent), it is lower than in the bottom (5.1 per cent) and second decile group (5.7 per cent). Furthermore, at each income level shown in Figure 32, there are households losing substantially more than 5 per cent of their income in relative earnings terms and others losing less, or gaining. In particular pensioners, where fully protected by earnings indexation, will see their incomes rising somewhat faster than the poverty line.

References

Adema, W. (2006) *Social Assistance Policy Development and the Provision of a Decent Level of Income in Selected OECD Countries*. OECD Social, Employment and Migration Working Papers No. 38. Paris: OECD

Bradshaw, J. and Lynes, T. (1995) *Benefit Uprating Policy and Living Standards*. Social Policy Reports No. 1. York: Social Policy Research Unit

Bradshaw, J. and Mayhew, E. (2006) 'Child benefit packages', in J. Bradshaw and A. Hartland (eds) *Social Policy, Family Change and Employment in Comparative Perspective.* Cheltenham: Edward Elgar

Brewer, M., Clark, T. and Goodman, A. (2003) 'What really happened to child poverty in the UK under Labour's first term?', *Economic Journal*, Vol. 113, No. 488, pp. F240–57

Brewer, M., Browne, J. and Sutherland, H. (2006a) *Microsimulating Child Poverty in 2010 and 2020*. York: Joseph Rowntree Foundation, www.jrf.org.uk/bookshop/details. asp?pubID=810

Brewer, M., Duncan, A., Shephard, A. and Suárez, M. (2006b) 'Did Working Families' Tax Credit work? The impact of in-work support on labour supply in Great Britain', *Labour Economics*, Vol. 13, No. 3, pp. 699–720

Brewer, M., Goodman, A., Shaw, J. and Sibieta, L. (2006c) *Poverty and Inequality in the UK: 2006.* IFS Commentary C101. London: Institute for Fiscal Studies

Brewer, M., Goodman, A., Muriel, A. and Sibieta, L. (2007) *Poverty and Inequality in the UK: 2007.* IFS Briefing Note BN73. London: Institute for Fiscal Studies

Buck, N., Sutherland, H. and Zantomio, F. (2007) *Tackling Child Poverty in London: Implications of Demographic and Economic Change*. London: Child Poverty Commission, http://213.86.122.139/publications/iser-rpt-0207.jsp

Cantillon, B. and Van Mechelen, N. (2004) *De bodembescherming in 15 Europese landenin de periode 1992–2001*. Antwerpen: Centrum voor Sociaal Beleid Herman Deleeck

Disney, R. and Johnson, P. (eds) (2001) *Pension Systems and Retirement Incomes across OECD Countries*. Cheltenham: Edward Elgar

DWP (Department for Work and Pensions) (2006a) *Abstract of Statistics for Benefits, National Insurance Contributions and Indices of Prices and Earnings: 2006 Edition.* London: DWP, www.dwp.gov.uk/asd/asd1/abstract/Abstract2006.pdf

DWP (Department for Work and Pensions) (2006b) *Security in Retirement: Towards a New Pensions System*. London: The Stationery Office

DWP (Department for Work and Pensions) (2006c) *Tax Benefit Model Tables April 2006*. Newcastle: DWP Information Directorate

DWP (Department for Work and Pensions) (2007a) *Households Below Average Income 1995/96–2005/06*. London: DWP

DWP (Department for Work and Pensions) (2007b) *Income Related Benefits: Estimates of Take-up in 2005/2006.* London: DWP

European Commission (2005) *Mutual Information System on Social Protection in the EU and European Economic Area*. Brussels: Directorate-General for Employment, Social Affairs and Equal Opportunities

Evans, M. and Eyre, J. (2004) *The Opportunities of a Lifetime: Model Lifetime Analysis of Current British Social Policy*. Bristol: Policy Press

Evans, M. and Scarborough, J. (2006) *Can Current Policy End Child Poverty by 2020?* York: Joseph Rowntree Foundation

Glennerster, H., Hills, J., Piachaud, D. and Webb, J. (2004) *One Hundred Years of Poverty and Policy*. York: Joseph Rowntree Foundation

Government Actuary's Department (2005) *Projections Database: 2004-based Principal Projections for the UK*. London: Government Actuary's Department, www. gad.gov.uk/Demography_Data/Population/Index.asp?y=2004&v=Principal&SubYear= Proceed

Gutierrez, R., Immervoll, H. and Sutherland, H. (2005) 'How European Union member states adjust tax and benefit systems for inflation', mimeo, www.iser.essex. ac.uk/msu/emod/micresa/docs/DL4c_InflationAdjustments.pdf

Hills, J. (2004) Inequality and the State. Oxford: Oxford University Press

Hills, J. and Sutherland, H. (2004) 'Ending child poverty in a generation? Policies and prospects in the UK', mimeo, presented at the conference on Cross-national Comparisons of Expenditures on Children, Princeton University, NJ, January

Hirsch, D. (2006) *What Would it Take to End Child Poverty? Firing on all Cylinders.* York: Joseph Rowntree Foundation

HM Revenue & Customs (2007) *Child Tax Credit and Working Tax Credit Take-up Rates 2004–05.* London: HMRC

HM Treasury (2007a) *Budget 2007: Economic and Fiscal Strategy Report*. HC 342. London: The Stationery Office, www.hm-treasury.gov.uk/budget/budget_07/bud_ bud07_index.cfm

HM Treasury (2007b) *Meeting the Aspirations of the British People: 2007 Pre-Budget Report and Comprehensive Spending Review*. CM7227. London: The Stationery Office

Johnson, P. (1996) *Sharing the Prosperity of the Nation: Uprating Pensions*. London: Age Concern England

Meade Committee (1978) *The Structure and Reform of Direct Taxation. Report of a Committee Chaired by Professor J.E. Meade.* London: George Allen & Unwin for the Institute for Fiscal Studies

OECD (2004) Benefits and Wages. Paris: OECD

OECD (2005) Pensions at a Glance. Paris: OECD

Pensions Commission (2006) *Implementing an Integrated Package of Pension Reforms. The Final Report of the Pensions Commission*. London: Pensions Commission, www.pensionscommission.org.uk

Redmond, G., Sutherland, H. and Wilson, M. (1998) *The Arithmetic of Tax and Social Security Reform: A User's Guide to Microsimulation Methods and Analysis.* Cambridge: Cambridge University Press Sutherland, H. (2004) 'Poverty in Britain: the impact of government policy since 1997. A projection to 2004–5 using microsimulation', Microsimulation Research Note MU/ RN/44, www.iser.essex.ac.uk/msu/publications/pdf/murn44.pdf

Sutherland, H., Sefton, T. and Piachaud, D. (2003) *Poverty in Britain: The Impact of Government Policy since 1997*. York: Joseph Rowntree Foundation

Appendix 1: Current benefit uprating policies¹

Section 150 of the Social Security Administration Act 1992 requires the Secretary of State to review the level of benefits annually to determine whether they have retained their value relative to the general level of prices. The Act provides that certain benefits must be uprated in line with prices. The Secretary of State considers the Retail Prices Index (RPI) to be the appropriate measure for uprating these benefits. Although the Secretary of State must review the level of income-related benefits, he is not required to increase them in line with prices. Historically, he has exercised his discretion to provide an increase and considers the most appropriate index for these benefits to be the RPI less costs that are met through the income-related benefits such as housing costs.

Benefits with statutory requirement for uprating

- Attendance Allowance
- Child Special Allowance
- Disability Living Allowance
- Industrial Death Benefits (existing cases only)
- Industrial Injuries Disablement Benefit
- Carer's Allowance
- Incapacity Benefit
- State Pension (including SERPS and Graduated Pension)
- Severe Disablement Allowance
- Widowed Mother's/Parent's Allowance
- Widow's Pension (including child and dependency additions) replaced by Bereavement Benefit

Benefits without a statutory requirement for uprating

- Child Benefit
- Child Tax Credit
- Council Tax Benefit*
- Housing Benefit*
- Income Support*
- Jobseeker's Allowance
- Maternity Allowance
- Statutory Maternity Pay
- Statutory Sick Pay
- Working Tax Credit
- * These benefits are not uprated by the RPI (all items). They are uprated by New Rossi, which is currently defined as RPI (all items) less rent, local taxes and mortgage interest payments.

Note

1. This appendix is based on an extract from *The Abstract of Statistics for Benefits, National Insurance Contributions and Indices of Prices and Earnings: 2006 Edition*, 'Appendix A: Uprating policy' (DWP, 2006a).

Appendix 2: Base case uprating assumptions

Table A2.1 shows how each element of the tax and benefit system is uprated according to the base case under the application of current uprating policies and assumptions. The values of the uprating factors are shown in relation to prices and earnings. We have assumed that the Retail Prices Index (RPI) and the Rossi index move up to 2011/12 as given in the 2007 Budget Report (HM Treasury, 2007a, Table C3). These indices apply to uprating of taxes and benefits in the following year. So, for example, the 2011/12 index will be used for uprating taxes and benefits in April 2013.¹ For the remaining 13 years of uprating (to 2026/27), we assume the same constant rate growth in RPI and Rossi as given for the final year, 2011/12 – 2.75 per cent for the RPI and 2.25 per cent for Rossi. Table A2.2 shows the nominal value of the indices used as the basis for the uprating of individual elements shown in Table A2.1.

Earnings uprating uses the change in Average Earnings up to the previous July. The specific index used in practice is the AEI LNNC.² However, there are no official projections or assumptions about how this will move in the future. We have made the assumption that real earnings growth is 2 per cent per year over the whole period. The resulting nominal index is shown in Table A2.2.

Table A2.1 Base case uprating assumptions					
	Current uprating policy or	Change relativ (RPI	e to prices) (%)	Change relativ (%	/e to earnings (₆)
Tax and benefit levels and thresholds	assumption	Six years	20 years	Six years	20 years
Income tax personal allowances and Married Couple's allowance	In line with RPI	0.00	0.00	-11.20	-32.70
Income tax thresholds and income limit for age-related allowance					
National Insurance contribution lower and upper limits and thresholds					
Child Benefit, Widowed Parent's Allowance, Bereavement					
Allowance, Contributory JSA, Incapacity Benefit, Carer's Allowance, Attendance Allowance, Severe Disablement					
Allowance, Disability Living Allowance, maternity pay and allowances. War Pensions					
All Working Tax Credit payments, additions and disregards Severe disability premiums on IS, HB, CTB					
Basic State Pension (BSP)	In line with RPI (with 2.5% minimum nominal increase)	2.00	34.59	-9.43	-9.43
	until year 7 (2012/13), then with earnings				
Family element of the Child Tax Credit, baby element of the Child Tax Credit, Child and Working Tax Credit thresholds	Frozen: no change in nominal amount	l –16.25	-42.71	-25.63	-61.45
Childcare element of the WTC maximum amounts Housing Benefit, Income Support, Council Tax Benefit and					
Pension Credit capital limits Earnings and other disregards in IS, HB and CTB Winter fuel payments to pensioners					
-					

Table A2.1 Base case uprating assumptions – Continued

	Current uprating policy or	Change relativ	re to prices	Change relativ	e to earnings
Tax and benefit levels and thresholds	assumption	Six years	20 years	Six years	20 years
Income Support, income-based JSA, Housing Benefit and Council Tax Benefit applicable amounts (except for pensioners and children) Non-dependent deductions and income limits in IS, HB, CTB Council Tax second adult rebate income thresholds	In line with Rossi index: 2.25 per year	-2.88	-9.29	-13.76	-38.95
Pension Credit Guarantee amounts (GC), income tax age-related allowances	In line with earnings	12.62	48.59	0.00	0.00
Savings Credit (SC) lower threshold Single	In line with RPI until year 3 (2008/09), then with earnings	10.41	57.41	-1.96	5.94
Couple	calculated from GC level and maximum payment (indexed by prices)	10.41	55.46	-1.96	4.62
Savings Credit (SC) maximum amount Single	According to a formula depending on the movement of the SC lower threshold and	18.86	23.66	5.54	-16.78
Couple	GC amounts until year 9 (2014/15) and then in line with RPI	20.19	25.04	6.72	-15.85
Per child element of the Child Tax Credit	In line with earnings until year 4 (2009/10) and then in line with RPI	6.12	6.12	-5.77	-28.58
Minimum wage, Council Tax	In line with earnings (by assumption)	12.62	48.59	0.00	0.00
Many rules for the uprating of benefit amounts involve round increase in tax thresholds up to the nearest $\pounds100$ a year and in the uprating of the systems shown here.	lding assumptions. Typically, ben Id the increase in tax allowances	nefits are round s to the neares!	ed to the neal t £10 a year. T	est 5 pence a w hese convention	eek, the is are ignored

The impact of benefit and tax uprating on incomes and poverty

Uprating applied in	2007/08	2008/09	2009/10	2010/11	2011/12	2012/13	2013/14 2026/27
RPIª	3.5	3.5	2.75	2.75	2.75	2.75	2.75
Rossi ^a	3.0	2.25	2.5	2.5	2.5	2.25	2.25
Earnings ^b	5.57	5.57	4.805	4.805	4.805	4.805	4.805

Table A2.2 Assumptions about year-to-year percentage changes in prices andearnings

All changes from last year, September–September for RPI, July–July for earnings (so 2006/07 uses September 2004–September 2005 and July 2004–July 2005).

a. HM Treasury (2007a, Table C3).

b. RPI with 2 per cent earnings growth.

Notes

- 1. The index of annual change up to the previous September is used and this is what is assumed by HM Treasury in the price indices shown in Table A2.2.
- 2. www.statistics.gov.uk/StatBase/tsdataset.asp?vlnk=392&Pos=1&ColRank=1&Ran k=-1.

Appendix 3: Structural changes announced in the 2007 Budget and accounted for in the 2006–07 starting point system

Budget announcement	What is modelled	Actual 2006–07 value	starting point value
Working and Child Tax Credits (WTC/CTC) taper increased to 39% in 2008	WTC/CTC taper increased to 39%	37%	39%
Basic rate of tax reduced to 20% in 2008	Basic rate of tax reduced to 20%	22%	20%
Reduced tax rate abolished in 2008	Reduced tax rate abolished	10%	20%
WTC/CTC threshold increased by £1,200 p.a. in 2008	WTC/CTC threshold increased by £1,200 (default is to freeze at nominal value)	£5,220	£6,420
Child Tax Credit (CTC) child amount increased by £150 p.a. over price indexation in 2008	CTC child amount increased by £150 less two years of earnings indexation (the assumed default); Housing Benefit and Council Tax Benefit child amounts increased accordingly	£1,765	£1,905
National Insurance contribution (NIC) Upper Earnings/Profits Limits (UEL) increased by £75 p.w. nominal in 2008 and to match effective Higher Rate Threshold ^a for tax (HRT) in 2009	NIC UELs increased to HRT in 2009 less three years of price indexation	£645 p.w. (£33,540 p.a.)	£751.19 p.w. (£39,062 p.a.)
HRT increased by £800 more than inflation in 2009	Tax upper threshold indexed in 2008 and 2009, then £800 less three years' price indexation added	£33,300 p.a.	£34,027 p.a.
Tax age allowances increased by £1,180 over price indexation in 2008; age 75+ allowance to reach £10,000 by 2011	Age allowances increased by $\pounds1,180$ less two years of price indexation; for 75+ allowance, $\pounds86$ (the amount added to reach $\pounds10,000$ in 2011) less five years of price indexation is also added	<75 £7,280; 75+ £7,420	<75 £8,381; 75+ £8,605
Child Benefit for the first child to reach £20 in 2010 (22p more than price indexation)	Increase first child premium by 22p less four years' price indexation (HB and CTB family premiums increased accordingly)	£5.75 p.w.	£5.95 p.w.

a. The effective HRT is the value of the upper threshold plus the personal allowance.

Appendix 4: Methods and models

POLIMOD

POLIMOD is a tax-benefit microsimulation model based on the UK Family Resources Survey (FRS).¹ All households in the sample are used. The data used in this analysis were collected in 2003–04 and the income variables are updated to 2006–07 levels of prices and incomes.² The household income variables used here to measure poverty among other things have been deliberately defined to be as similar as possible to those used in the Households Below Average Income (HBAI) statistics (DWP, 2007a). There are some minor departures from HBAI methodology due to the fact that POLIMOD simulates taxes and benefits in order to evaluate changes in the rules that govern them.

POLIMOD calculates liabilities for income tax and National Insurance contributions (NICs) and entitlements to Child Benefit, Working Tax Credit (WTC), Child Tax Credit (CTC), Income Support (IS) – including income-related Jobseeker's Allowance – Pension Credit (PC) including the Savings Credit (SC), Housing Benefit (HB) and Council Tax Benefit (CTB). Otherwise, elements of income are drawn from the recorded values in the FRS dataset.

As in HBAI analysis, incomes are measured before housing costs (BHC) and after housing costs (AHC). BHC income includes all original incomes (including private pensions) and all benefits (including public pensions and HB) and tax credits, less income tax and employee and self-employed NICs. AHC income is BHC income less net housing costs (rent, mortgage interest less HB plus any other housing payments).

POLIMOD captures the effects of non-take-up of means-tested benefits and tax credits (CTC, WTC, IS, PC, SC, HB and CTB) by applying the take-up proportions estimated on a caseload basis by the Department for Work and Pensions (DWP, 2007b) and HM Revenue & Customs (2007).³ For example, we assume that some 7 per cent of lone parents do not receive the combination of CTC and WTC to which they are entitled, and that this proportion is higher in London (37 per cent of all families with children). Of those entitled to PC, 25 per cent do not take up, with the proportion much higher (55 per cent) if there is only entitlement to SC. In general, we assume that take-up behaviour is not affected by changes in the size of benefit or tax credit entitlements.

In common with HBAI, the POLIMOD analysis uses the OECD equivalence scale to adjust BHC incomes for differences in household size and composition. AHC incomes are adjusted by the so-called companion scale. The scale values are 1 for a couple and 0.20 for a child under 14. A single person is weighted by 0.67 for BHC incomes and 0.58 for AHC incomes, and additional adults 0.33 and 0.42 respectively.

CARESIM and technical details underlying the analysis in Chapter 10

CARESIM is a microsimulation model whose main function is to analyse the distributional effects of alternative long-term care-charging regimes. It also simulates older people's income tax liability and entitlement to means-tested benefits under alternative policy options and it is this feature that is used here. CARESIM is based on a sample of older people drawn from the UK Family Resources Survey. It performs simulations for a base year and for future years. The initial sample consists of people aged 65 and over whose partner, if they have one, has reached state pension age. This sample is 'aged' for this report by six and 20 years. This involves the following.

- Predicting whether each member of the initial sample will be alive in six and 20 years' time. This uses the age and gender-specific survival probabilities underlying the Government Actuary's population projections.
- Predicting whether each sample member will have become a widow or widower in six and 20 years' time. This is done once survival of each partner has been predicted.
- Simulating how the income and wealth of each sample member and partner will have changed after six and 20 years, on the assumptions stated in the main text.
- Modelling any inheritance of income and capital of the surviving partner, if there is one.

Assumptions are the following.

- The survivor inherits all of his/her late partner's financial assets and income from them.
- The survivor inherits half of any non-state pension income that his/her partner had.

- The survivor inherits a proportion of the late partner's State Earnings Related Pension (SERPS) income, which falls from 100 per cent if the partner reached state pension age in 2002 or earlier, to 50 per cent if he/she attains state pension age after 2010, in line with announced government policy.
- If the survivor is a woman and her own Basic State Pension is less than that of her late husband's, her own Basic State Pension is increased by the difference between the two.

No allowance is made for changes in household composition except as a result of the death of a partner.

The sample is not 'refreshed', i.e. as individuals in the original sample are aged, no new individuals are added to replace those at younger ages. By 20 years from the base year, the simulations are therefore representative only of people aged 85 and over.

Non-take-up of means-tested benefits is allowed for in the same way as in POLIMOD.

Tax-benefit models for hypothetical model families

The calculation of taxes and benefits for hypothetical families in Chapters 6 and 7 adopts the methodology and definitions employed by the Department for Work and Pensions' Tax Benefit Model Tables (TBMT) (DWP, 2006c). The calculations reported here have been adapted from the original spreadsheets to calculate taxes and benefits using incremental increases in hours worked at the National Minimum Wage for an adult (£5.35 per hour in October 2006), rather than bands of gross earnings used in TBMT. We also change the DWP assumptions on the calculation of before housing costs income in TBMT, deducting Council Tax as an element of tax, rather than treating it as a housing cost and thus as a deduction only from AHC income. This change ensures consistency with both POLIMOD and DWP's Households Below Average Income series. The assumptions for rent and Council Tax levels follow the TBMT assumptions of changing amounts to reflect family size and tenure. All calculations in this report adopt the assumption of social sector rents, termed 'LA rents' in TBMT.

The starting point model adopts the changes introduced in the 2007 Budget based on the 2006 system. This makes use of the same policy parameters as POLIMOD, described above, and also adopts the poverty lines based on 60 per cent of median income derived from POLIMOD. The models that describe the tax-benefit systems after six and 20 years of base case uprating also adopt the same tax and benefit parameters as POLIMOD, and make use of the poverty lines generated by POLIMOD under base case uprating assumptions.

Lifetime Opportunities and Incentives Simulation (LOIS)

LOIS uses the same underlying methodology as the cross-sectional tax benefit models for hypothetical families discussed above, except that the profile is longitudinal. This means that changes are analysed between different points of time, rather than solely according to potential changes in concurrent earnings. Adding the dimension of time means that tax-benefit calculations over time for a hypothetical family reflect a combination of lifetime events and trends, and their interaction with tax and benefit policy.

Two reduced forms of lifetime profiles are used in this report to reflect the 20-year projections used in Chapters 6 to 9. The first set of profiles, used in Chapter 9, creates 'model childhoods' for 20 years for a couple who have their first baby in 2006 followed by a second child three years later. These birth profiles allow a continuous period of 'childhood' to be profiled over 20 years over the combination of two children. The first, however, ceases to be a child after 18 years. The parents of these children are given both earnings histories and wage rates.

The second LOIS model, used in Chapter 10, illustrates the position of hypothetical people who retire in 2006 aged 65. The profiles employ a simplifying assumption that couple members retire together at age 65. To focus on the effects of uprating, no demographic events are included in these illustrative 'model retirements' – put simply, the retiree couple get older together and face neither death nor separation.

LOIS analysis in Chapters 9 and 10 adopts the baseline from current uprating practice outlined in Chapters 5 and 6, and all policy parameters and poverty lines are aligned with those used in, or generated by, POLIMOD. This means that the individual profiles from each policy alternative can be compared consistently to poverty lines that reflect that alternative across the whole population.

Notes

- 1. Department for Work and Pensions, 2003–04 Family Resources Survey. Distributed by the Economic and Social Data Service. Crown Copyright material is reproduced with the permission of the Controller of HMSO and the Queen's Printer for Scotland.
- 2. See Redmond et al. (1998) for more information.
- 3. Where ranges of take-up proportions are published, the mid-point is used.